ELECTROCATALYTIC WATER SPLITTING TO PRODUCE FUEL HYDROGEN

By

HAO YUAN

A DISSERTATION

Submitted to Michigan State University in partial fulfillment of the requirements for the degree of

Chemical Engineering – Doctor of Philosophy

ABSTRACT

ELECTROCATALYTIC WATER SPLITTING TO PRODUCE FUEL HYDROGEN By

HAO YUAN

Solar energy is regarded as a promising source for clean and sustainable energy. However, it is not a continuous energy source, thus certain strategies have to be developed to effectively convert and store it. Solar-driven electrocatalytic water splitting, which converts solar energy into chemical energy for storage as fuel hydrogen, can effectively mitigate the intermittence of solar radiation. Water splitting consists of two half reactions: water oxidation and hydrogen evolution. Both reactions rely on highly effective electrocatalysts. This dissertation is an account of four detailed studies on developing highly effective low-cost electrocatalysts for both reactions, and includes a preliminary attempt at system integration to build a functional photoanode for solar-driven water oxidation.

For the water oxidation reaction, we have developed an electrochemical method to immobilize a cobalt-based (Co-OXO) water oxidation catalyst on a conductive surface to promote recyclability and reusability without affecting functionality. We have also developed a method to synthesize a manganese-based (MnO_x) catalytic film *in situ*, generating a nanoscale fibrous morphology that provides steady and excellent water oxidation performance. The new method involves two series of cyclic voltammetry (CV) over different potential ranges, followed by calcination to increase crystallinity. The research has the potential to open avenues for synthesizing and optimizing other manganese-based water oxidation catalysts.

For the hydrogen evolution reaction, we have developed a new electrodeposition method to synthesize Ni/Ni(OH)₂ catalysts in situ on conductive surfaces. The new method involves only two cycles of CV over a single potential range. The resulting catalytic film has a morphology of packed walnut-shaped particles. It has superior catalytic activity and good stability over long periods.

We have inestigated the feasibility of incorporating manganese-based water oxidation catalysts into TiO_2 to build a composite photoanode. Results show that the composite photoanode has good activity under ultraviolet (UV) illumination, and that its catalytic performance can be significantly improved by enabling light collection over a wider range of wavelengths. We have also proposed a scheme for a solar-driven water splitting device that integrates both electrocatalysts with solar energy collection and intensification, and is capable of minimizing mass transfer resistance. The dissertation concludes with suggested future work to further explore the MnO_x and Ni/Ni(OH)₂ electrocatalysts. Copyright by HAO YUAN 2017 To my parents, and all of my teachers and professors who have been outstanding mentors throughout my entire academic career.

ACKNOWLEDGEMENTS

I would like to first thank my advisor, Dr. Robert Y. Ofoli, for being an excellent mentor and encouraging my research at all times. His advice on both research and my career has been invaluable.

I would also like to thank my tremendous committee members, Dr. Richard R. Lunt, Dr. Gary J. Blanchard and Dr. Ilsoon Lee for serving on my committee and providing brilliant suggestions and great help throughout my doctoral work at MSU. My thanks also to Dr. Yanzong Zhang, who first led me into the world of science.

Next, I would like to thank everyone who has helped me with my research: Dr. Fleglers, Dr. Fan, Amy, Dr. Staples, Dr. Askeland, and all the staff personnel in the CHEMS department.

Big thanks to my all my dear friends for all the good times we enjoyed together as well as the times we wasted together. Special thanks to my roommate Weijie Ren for his great friendship and help, and heartfelt thanks to my beloved girlfriend, Xiaolei, for her years-long support and encouragement.

Finally, I would like to thank my family, especially my parents, for all the support and sacrifices that they have made for me throughout my life.

TABLE OF CONTENTS

LIST OF TABLES	X
LIST OF FIGURES	xi
KEY TO ABBREVIATIONS	xvii
CHAPTER 1. INTRODUCTION	1
1.1 Overview and significance of the problem	1
1.2 Natural photosynthesis and artificial photosynthesis (AP) schemes	5
1.3 Outline of this dissertation	7
CHAPTER 2. CHARACTERIZATION AND FUNCTIONAL ASSESSMENT OF A	
COBALT(III)-OXO CUBANE CLUSTER WATER OXIDATION CATALYST	
IMMOBILIZED ON A CONDUCTIVE SURFACE	9
21 Abstract	9
2.2 Introduction	9
2.2 Materials and methods	12
2.3 1 Materials	
2.3.1 Materials	12
2.3.2 Treating situ Synthesis of the Co-OXO cubane cluster	12
2.3.2.7 Ex studies of the Co OXO cubane cluster information $2.3.2.7$ Electrodenosition of Co-OXO catalyst synthesized ex situ	12
2.3.2.2 Electrodeposition of Co ONO cutaryst synthesized on situ	13
2.3.2.5 Cutarytic functionality and stability assessments	
2.5.2.1 Characterization of catalysts by 1 The and SERT	14
2.4.1 Flectrodenosition of Co-OXO	14
2.42 FTIR and surface FTIR characterization of the Co-OXO catalyst	16
2.4.2 Surface morphology and structure characterizations of immobilized catalyst	10
2.4.5 Surface morphology and structure characterizations of miniobilized catalyst	17 19
2.5 Conclusions	
CHARTER 2 GANTHERIC OF A M. O. CATALATIC FILM ON FTO DX DUAL CED	IEC
CHAPTER 5.51NTHESIS OF A MIIOX CATALTTIC FILM ON FTO DT DUAL-SER	
2.1 Abstract	
3.1 Abstract	
3.2 Introduction	
3.3 Materials and methods	
3.3.1 Materials	
3.3.2 Electrodeposition of MnO _x on F1O	
3.3.3 Surface morphology characterization	
3.3.4 Catalyst functionality assessments and comparisons	
3.3.5 Catalyst stability assessment	
3.4 Kesults and discussions	
3.4.1 Surface morphology characterizations	
3.4.2 Catalytic functionality assessments	32

3.4.3 Proposed formation mechanism	35
3.4.4 XRD characterizations	41
3.4.5 Stability tests	43
3.5 Conclusions	47
CHAPTER 4. FUNCTIONAL BIOMIMETIC COMPOSITE Mn/TiO2 PHOTOANODE FO	OR
WATER SPLITTING	48
4.1 Abstract	48
4.2 Introduction	48
4.3 Materials and methods	50
4.3.1 Materials	50
4.3.2 Methods	51
4.2.2.1 Spin coating of the TiO ₂ film	51
4.3.2.2 In situ synthesis of the Mn WOC catalyst	51
4.3.2.3 Structural characterization by XRD and SEM	52
4.3.2.4 Functionality assessment by CV and EIS	52
4.4 Results and discussion	53
4.4.1 Structural and morphological characterization of the composite photoanode	53
4.4.2 Electrochemical characterization of the Mn WOC	58
4.4.3 Electrochemical characterization of the composite photoanode.	59
4.5 Conclusions	63
CHAPTER 5. ELECTRODEPOSITION OF Ni/Ni(OH) ₂ CATALYTIC FILMS BY CYCL	IC
VOLTAVINIETRY FOR THE HYDROGEN EVOLUTION REACTION	04
5.1 Adstract	04
5.2 Introduction	04
5.5 Experimental session	03
5.3.1 Waterials	03
5.3.2 Synthesis by electrodeposition	00
5.3.5 Waterials characterization.	00
5.4 Decults and discussions	07
5.4 1 Structural characterization	00
5.4.1 Structural characterization	08
5.4.2 Functionality assessments	12
5.4.5 Stability assessments	70 77
5.4.4 Effects of electrolyte concentration on electrodeposition	//
5.4.5 Hypothesis of the formation mechanism	70
5.5 Conclusions	79
CHAPTER 6. SUGGESTIONS FOR FUTURE RESEARCH	80
6.1 Introduction	80
6.2 Further research on improving the MnO _x WOC	82
6.3 Further research on improving the Ni/Ni(OH) ₂ HER catalyst	82
6.4 Further research on system integration	83
	'
CHAPTER 7. CONCLUSIONS	86

LIST OF TABLES

Table 5.1. Comparison of Tafel slopes and exchange current densities of cc-Ni, cv-Ni, and Ni foil catalysts assessed in neutral media. The lower slope and higher exchange current density of cy-Ni
are indicators of much better catalytic performance
Table 5.2. Peak current density at -0.5 V vs. NHE for samples synthesized in different concentrations of electrolyte

LIST OF FIGURES

Figure 1.1. Diagram showing our current energy demand. Currently, we utilize 17.3TW of energy, with more than 70% provided by fossil fuels. Solar energy only contributes ~10%......3

Figure 2.4. Cyclic voltammograms acquired at a scan rate of 10 mV/s on bare ITO (blue curve), Co-OXO immobilized on gold (red), and Co-OXO immobilized on ITO (green) in 0.1M neutral phosphate buffer solution at pH 7.0. This diagram shows that water oxidation occurs at approximately 1 V for Co-OXO on both gold and ITO. For comparison, we have also included the CV of the catalyst in a pseudo-homogeneous system (purple). The profiles of the three scans are

essentially identical, providing strong evidence that Co-OXO maintains its catalytic function whether in solution, or immobilized on gold or ITO......20

Figure 2.6. Assessment of catalyst recyclability and reusability. In this test, we acquired eight CV scans on a single sample of Co-OXO catalyst coated on ITO. After each scan, the sample was taken out of the solution, rinsed with DI water, dried in air and reinserted into the solution for the next scan. As the data show, all the voltammograms are distributed over a small range of values, confirming a high degree of catalyst reusability, as well as reproducibility in performance.....23

Figure 3.6 b. SEM images at various magnifications of sample after four (4) cycles in the first CV series. Small clusters in Figure 3.6 a have grown into a larger scale......37

Figure 3.8. X-ray diffraction spectra of the three catalysts: a) MnO_x after two consecutive series of CV electrodeposition (scans over 0-0.6 V and 0.6-2.0 V) followed by calcination; it has the peaks of both Mn_2O_3 and MnO_2 , which indicates that the MnO(OH) formed in the first CV series has been further oxidized. b) MnO_x after one series of CV (0-2.0 V) and calcination, showing a similar pattern to trace a); this is an indication that the single 0-2.0 V series produces a MnO_x catalyst with a similar oxidation state as the one synthesized by the dual-series. c) MnO_x after one CV series (0-0.6 V) and calcination; the scan shows that the major phase of MnO_x is Mn_2O_3 , meaning that the MnO_x before calcination is MnO(OH) as proposed. The symbols +, *, and # in this figure designate the feature peaks of the FTO substrate, Mn_2O_3 and MnO_2 , respectively......42

Figure 3.9. Current density profile of the MnO_x catalyst synthesized by the dual-series method during long term catalytic assessment at 1.2 V (vs. SHE) in a 0.1 M phosphate buffer solution (pH 7.0). The steady profile over a long period of time gives an indication of a stable MnO_x film.

Figure 6.1. Schematics of two water splitting devices. Each scheme has a photoanode complex containing a solar harvester and a WOC, and a counter electrode for hydrogen ion reconstitution to produce fuel. a. Two-chip protocol (face-to-face scheme), where the counter electrode does not

KEY TO ABBREVIATIONS

AP	artificial photosynthesis
CV	cyclic voltammetry
DSSC	dye sensitized solar cell
EDS	dispersive x-ray spectroscopy
EIS	electrochemical impedance spectroscopy
EXAFS	extended x-ray absorption fine structure
FESEM	field emission scanning electron microscope
FTIR	Fourier transform infrared spectroscopy
FTO	fluorine-doped tin oxide
HER	hydrogen evolution reaction
HERC	hydrogen evolution reaction catalyst
ΙΤΟ	indium tin oxide
NADPH	nicotinamide adenine dinucleotide phosphate
NHE	normal hydrogen electrode
OEC	oxygen evolution complex
PBS	phosphate buffer solution
PV	photovoltaic
SEM	scanning electron microscope
TEM	transmission electron microscope
UV	ultraviolet
WOC	water oxidation catalyst

XANES x-ray absorption near edge structure

- **XAS** x-ray absorption spectroscopy
- **XRD** x-ray powder diffraction

CHAPTER 1. INTRODUCTION

1.1 Overview and significance of the problem

Society relies on sufficient energy supplies to meet a variety of needs¹. Currently, more than 70% of the global energy supply comes from fossil fuels, including crude oil, natural gas and coal (Figure 1.1). However, the rate of consumption of fossil fuels is much higher than the carbon fixation rate of natural photosynthesis². Also, due to its uneven distribution among countries, decreasing availability, and green-house gas emission issues, fossil fuels are no longer an ideal energy source. This has created a critical global challenge for sustainable energy production. The most effective way to do this is to produce energy from renewable and/or earth-abundant materials.

Solar energy is the only renewable energy source that is capable of powering the entire planet. It has a net output of 125,000 TJ/s, which is magnitudes higher than our current global energy needs^{3,4}. However, there are two major drawbacks that prevent solar energy from practical utilization on a larger scale. The first drawback is that, except in very few situations, it cannot be used directly⁵. It always needs to be converted into other forms such as heat⁶, electrical^{7.9} or chemical energy¹⁰⁻¹³. Another drawback is that solar radiation is not a continuous energy source. Sunlight is generally available during the day but not at night¹⁴. Thus, strategies have to be developed to store the energy collected during the periods of sunshine. Traditionally, this intermittency has been mitigated by photovoltaic (PV) technology, which stores solar energy in batteries, or solar-thermal technologies which store solar energy in the form of heat. Artificial photosynthesis (AP), a process that mimics natural photosynthesis, has been gaining attention

recently as a promising strategy to convert and store solar energy in the form of chemical energy¹⁵. Typically, there are three artificial photosynthesis schemes: direct hydrogen fuel production through water splitting, direct carbon-based fuel production, and indirect carbon-based fuel production. Figure 1.2a portrays the direct hydrogen production via water splitting, while Figures 1.2b and 1.2c are schemes for direct and indirect carbon-based fuels production that also require carbon dioxide (CO_2) sequestration. In each of the three schemes, the only remnants following consumption of the produced fuels are the water (H_2O) and carbon dioxide (CO_2) used to produce them. Thus, all three schemes are self-sustainable and carbon-neutral.

This dissertation focuses on our group's research on direct hydrogen fuel production via catalytic solar-driven water splitting (Figure 1a).



Figure 1.1. Diagram showing our current energy demand. Currently, we utilize 17.3TW of energy, with more than 70% provided by fossil fuels. Solar energy only contributes ~10%.



Figure 1.2. Three schemes of typical artificial photosynthesis processes: a) direct hydrogen fuel production through solar-driven water splitting; b) direct carbon-based fuel production using hydrogen ions that react directly with captured CO_2 ; c) indirect carbon-based fuel production with captured CO_2 reacting with reconstituted hydrogen gas.

1.2 Natural photosynthesis and artificial photosynthesis (AP) schemes

There are two main catalytic reactions in the natural photosynthesis process^{16,17}: the lightdependent reaction (light reaction); and the light-independent reaction (dark reaction). In the light reaction, pigments in chlorophyll capture sunlight and use the converted solar energy to oxidize water, producing oxygen gas and protons (in the form of NADPH)^{18,19}. In the following dark reaction, the protons (NADPH) are used to produce organic molecules with CO₂ through the Calvin cycle²⁰. A biomembrane separates the two processes, and provides a means to selectively transfer protons (NADPH).

Similar to natural photosynthesis, the solar-driven electrocatalytic water splitting process involves two half reactions²¹: the water oxidation reaction generates oxygen gas and hydrogen ions (Equation 1.1), and the hydrogen evolution reaction reconstitutes the hydrogen ions into hydrogen gas (Equation 1.2).

 $H_2 O \rightarrow \frac{1}{2} O_2 + 2H^+ + 2e^-$ (1.1) $2H^+ + 2e^- \rightarrow H_2$ (1.2)

As a result, a typical water splitting device has two major components: a solar-powered photoanode and a catalytic hydrogen evolution cathode. Shown in Figure 1.3, the photoanode usually consists of a robust substrate, a layer or layers of semiconducting material for solar energy capture, and a highly-effective water oxidation catalyst. The cathode side consists of a conductive substrate and a hydrogen evolution catalyst.



Figure 1.3. Typical solar-driven electrocatalytic water splitting device to produce hydrogen fuel. It usually consists of a composite photoanode and a counter electrode. The composite photoanode should have a robust conductive substrate, a solar harvester which can convert solar energy into the potential required to power the water splitting reaction, and a highly effective water oxidation catalyst to lower the overpotential of the water oxidation reaction. The counter electrode should also have a conductive substrate as well as a hydrogen evolution catalyst for hydrogen reconstitution.

1.3 Outline of this dissertation

The goal of this dissertation is to present the details of our work on developing electrocatalysts for both the water oxidation and hydrogen evolution reactions. We will also present a scheme for system integration to make functional photoanodes for solar-driven water oxidation.

Chapter 2 will focus on our research on the synthesis, immobilization and assessment of a cobaltbased (Co-OXO) water oxidation catalyst (WOC). We developed an electrodeposition method to successfully immobilize the Co-OXO WOC synthesized ex situ onto conductive surfaces. The resulting film enabled enhanced recyclability and reusability, while maintaining catalyst functionality in solution.

Chapter 3 describes our research on developing a dual-series CV electrodeposition method to synthesize a MnO_x WOC in situ on conductive surfaces. The resulting film provides excellent catalytic performance over long periods of time.

Chapter 4 presents our work on a preliminary system integration of TiO_2 and a Mn WOC to build a functional water oxidation photoanode for solar-driven water oxidation. Assessment of the photoanode showed that it is very effective towards water oxidation under UV illumination.

In Chapter 5, we describe our work on developing a Ni/Ni(OH)₂ electrocatalyst for the hydrogen evolution reaction. The new electrodeposition method involves only two cycles of CV in a single potential range, and produces a film with excellent catalytic properties.

Chapter 6 presents some suggestions for future research. We propose a device capable of minimizing mass transfer resistance, and also suggest future research activities on enhancing the performance of MnOx and Ni/Ni(OH)₂ electrocatalysts.

Chapter 7 presents the conclusions of the dissertation.

CHAPTER 2. CHARACTERIZATION AND FUNCTIONAL ASSESSMENT OF A COBALT(III)-OXO CUBANE CLUSTER WATER OXIDATION CATALYST IMMOBILIZED ON A CONDUCTIVE SURFACE

2.1 Abstract

We have developed an electrodeposition method that successfully immobilizes a cobalt(III)-OXO (Co-OXO) water oxidation catalyst (WOC) synthesized ex situ onto conductive surfaces. This Co-OXO WOC has a cubane-like catalytic core surrounded by stabilizing ligands, which is a structural mimic of the natural oxygen evolving complex (OEC). This method promotes recyclability and reusability of the Co-OXO catalyst. We took Fourier transform-infrared (FTIR) and surface FTIR spectra before and after deposition, respectively, to assess the presence of organic ligands around the core. We used scanning electron microscopy (SEM) to characterize the surface morphology of the immobilized catalyst, and energy dispersive x-ray spectroscopy (EDS) to determine the elemental composition. Finally, we assessed its functionality, stability, and reusability using cyclic voltammetry (CV). The results show that this catalyst can be successfully immobilized on conductive surfaces, achieving a functionality comparable to that of the same catalyst dissolved in aqueous solutions. We believe the separation of the synthesis and immobilization processes has great potential for optimizing catalytic perfomance.

2.2 Introduction

As discussed in Chapter 1, a complete water splitting process consists of two half reactions^{22,23}: water oxidation and hydrogen evolution. In this study, we focused mainly on the water oxidation half reaction. The water oxidation reaction is important in the natural process of photosynthesis, and the oxygen evolving complex (OEC) in photosynthetic organisms is the only known natural system capable of catalyzing that reaction. It has been shown that the natural OEC has a structure

consisting of an "oxo-bridged heteronuclear cubane cluster of manganese and calcium"²⁴. Much progress has been made in the past two decades towards synthesizing efficient WOCs that are structural mimics of the natural OEC. Since 2000, various catalysts with similar cubane structures have been synthesized and assessed. Successful examples include those developed by Kanady et al.²⁵, Mukherjee et al.²⁶, and Brimblecombe et al²⁷.

Dismukes and co-workers are generally recognized as the first group to successfully synthesize a WOC that structurally simulates the natural OEC^{28} . Their catalyst had a manganese cubane (Mn₄O₄) core surrounded by stabilizing ligands. A significant problem with this catalyst is that it is insoluble in water and common organic solvents²⁹. As a result, they used porous membranes such as Nafion to immobilize it for assessment and utilization. Other manganese-based compounds such as Mn₃Ca and Mn₃CaO₄ have also been developed recently for the same purpose. In 2007, Das and co-workers introduced a cobalt(III)-based WOC [Co4(µ3-O)4(µ- $O_2CMe_4(py)_4$], which is a complex with a [Co_4O_4] cubane-like catalytic core surrounded by pyridine and acetate anion stabilizing ligands³⁰. This complex is soluble in water. Thus, its electrochemical and catalytic properties can be easily assessed by a standard electrochemical system, with dissolved catalyst as electrolyte. In this chapter, we refer to this as the "pseudohomogeneous" system because, even though the catalyst is dissolved in water, the reaction takes place at the electrode-electrolyte interface. While the pseudo-homogeneous system works well on a laboratory scale and enables highly effective catalyst performance, it does not enable catalyst recovery and reuse on a larger scale. As a result, strategies have to be developed to immobilize the catalysts onto substrates to promote recyclability and reusability. The successful immobilization of these WOCs has scientific, technological and environmental merit.

In this study, we focused mainly on designs that promote catalyst recyclability and reusability. We have achieved this by developing a new electrodeposition protocol which cam immobilize the catalyst onto conductive surfaces. This method can transform the pseudo-homogeneous reaction system to a heterogeneous system. We chose a Co-OXO WOC as the candidate catalyst, with the goal of determining whether the OEC-mimicking catalytic core (with or without stabilizing ligands) can maintain catalytic function after being anchored onto conductive surfaces. We took Fourier transform-infrared (FTIR) and surface FTIR spectra before and after deposition, respectively, to investigate the presence of organic ligands around the core. We used scanning electron microscopy (SEM) to characterize the surface morphology of the immobilized catalyst, and energy dispersive x-ray spectroscopy (EDS) to determine the elemental composition. Finally, we assessed catalyst functionality, stability and reusability using cyclic voltammetry (CV). The results show that this catalyst can be successfully immobilized on conductive surfaces, with a functionality comparable to the same catalyst dissolved in aqueous solutions. Although some WOCs can be synthesized in situ, we still believe that the separation of catalyst synthesis and immobilization processes can enable better control of the catalyst and provide an objective route to optimization of the resulting complex.

2.3 Materials and methods.

2.3.1 Materials

Cobalt nitrate hexahydrate (CoCl₂•6H₂O), sodium acetate trihydrate (CH₃COONa•3H₂O), pyridine (C₅H₅N) and hydrogen peroxide (H₂O₂) were all purchased from the Sigma-Aldrich Company, LLC (St. Louis, MO). All chemicals were used without further purification. Polyethylene terephthalate (PET) films coated with indium tin oxide (ITO) were also purchased from Sigma-Aldrich. Prior to using, ITO substrates were cut into 2cm by 1cm rectangular chips.

2.3.2 Methods

2.3.2.1 Ex situ Synthesis of the Co-OXO cubane cluster

We used the protocol reported by Chakrabarty et al. (2007) to synthesize the $[Co_4(\mu_3-O)_4(\mu-O_2CMe)_4(py)_4]$ catalyst. In the *ex situ* synthesis, 20 mmol (4.76 g) of CoCl₂ was dissolved in 70ml methanol. The solution was kept at 70°C in a water bath with constant reflux. Forty (40) mmol (5.44 g) sodium acetate was added to the stirred solution, followed by addition of 16ml (10mmol) of pyridine (py). Ten (10) ml H₂O₂ (30% v/v) was then added very slowly. A very vigorous oxidation process could be observed, with the color of the mixture changing from purple to black. The mixture was maintained at 80°C for 4 hours. After three days of aging, a dark green powder was observed.

2.3.2.2 Electrodeposition of Co-OXO catalyst synthesized ex situ

The electrochemical deposition was carried out on a CHI 660D electrochemical workstation (CH Instruments Inc., Austin, Texas), using the traditional three-electrode setup. ITO chips were used as both the working and counter electrodes, with Ag/AgCl as the reference electrode. We dissolved 0.02 g of dry $[Co_4(\mu_3-O)_4(\mu-O_2CMe)_4(py)_4]$ powder in an electrolyte containing 0.1 M neutral phosphate buffer solution (NaH₂PO₄/Na₂HPO₄ buffer solution at pH 7.0). A potential of +1.2 V was applied to the system for seven (7) hours. For comparison, an identical experiment was run for three (3) hours, except without the catalyst in the solution.

2.3.2.3 Catalytic functionality and stability assessments

The electrodeposition process was tracked by CV, using a typical three-electrode setup on the CHI 660D electrochemical workstation. We did a baseline test using a blank ITO chip as the working electrode. We used two types of electrodes in the catalyst functionality tests: a Co-OXO-coated ITO electrode and a Co-OXO-coated gold electrode. A platinum wire was used as the counter electrode in all tests, with a Ag/AgCl electrode as reference, and the neutral 0.1M phosphate buffer solution (pH 7.0) as the electrolyte. The scan rate was set at 10 mV/s.

The functionality tests were also done using CV in an identical setup as described above. We tested the Co-OXO WOC coated on ITO and on gold substrates. Finally, we tested the bare ITO and pseudo-homogeneous systems for comparison. Following these assessments, we tested the stability and functionality of the catalyst. We obtained 14 CV scans on one sample of Co-OXO immobilized on ITO. In this test, the Co-OXO-coated ITO was used as the working electrode, and was left in the electrolyte solution for 14 consecutive CV scans. To test the recyclability of

the catalyst, we also took eight CV scans on one Co-OXO on ITO sample. Unlike the previous tests, the working electrode was removed from solution after each CV scan, rinsed in DI water, dried in air, and reinserted for the next scan.

2.3.2.4 Characterization of catalysts by FTIR and SEM

We used FTIR to evaluate the structure and morphology of the catalyst in powder form, and following immobilization on ITO. The data for the immobilized catalyst were acquired on a Nicolet 6700 surface FTIR (Thermo Scientific, Inc., West Palm Beach, FL). The data for the catalyst in powder form were acquired on a Galaxy Series 3000 spectrophotometer (Mattson Technology, Inc., Fremont, CA). SEM images were obtained at different magnifications using a JSM 6610-LV electron microscope (JEOL Ltd, Tokyo, Japan). Finally, energy dispersive x-ray spectroscopy (EDS) (Oxford Instruments, High Wycomb, Bucks, England) was used to evaluate the elemental composition of the immobilized catalysts.

2.4 Results and Discussion

2.4.1 Electrodeposition of Co-OXO

A dark film could be clearly observed after the electrodeposition process. We believe this provides direct evidence of the successful coating of the catalytic film on the conductive surface. A control experiment was also performed under identical conditions without dissolved Co-OXO in the electrolyte. The control sample did not produce any color changes on the surface. We recorded the current density profile during the seven-hour electrodeposition process. The red curve in Figure 2.1 shows that the current density of the catalyst sample increased steadily before reaching a steady state value. The blue curve in Figure 2.1 shows the current density profile of

the control sample, which decreased gradually, then reached and maintained a steady state at a current density level magnitudes lower than the electrodeposition sample. We should notice that there is an initial decrease of current density during the first few minutes of deposition. Although we do not have a clear explanation why this occurs, we speculate that it is due to mass transfer resistances³¹.



Figure 2.1. Current density profile (red) of the Co-OXO WOC during electrodeposition on ITO. The current density increased gradually to a maximum of 0.67 mA/cm², which is considerably higher than for the case where the electrode was placed in solution without the catalyst (blue curve). Similar results were obtained on gold electrodes. The initial drop in current density is likely due to mass transport effects.

2.4.2 FTIR and surface FTIR characterization of the Co-OXO catalyst

The FTIR spectrum of the catalyst powder sample synthesized *ex situ* has strong peaks at frequencies of 1556, 1407 and 669 cm⁻¹ (Figure 2.2), which were assigned to the pyridine ring group, the acetate hydroxyl group, and the carbon backbone -CH- group, respectively. These peaks confirmed the existence of the ligands used to stabilize the catalytic complex during the *ex situ* synthesis process. However, those feature peaks disappeared on the surface FTIR spectrum of the immobilized catalyst samples, which indicates the loss of all organic ligands during electrodeposition, leaving only the catalytic cubane core on the surface.



Figure 2.2. FTIR spectra of the catalyst powder prior to immobilization on conductive surfaces. The stretching frequencies at 1556, 1407, and 669 cm⁻¹ represent the pyridine ring group, the acetate hydroxyl group, and the carbon backbone -CH- groups, respectively. These frequencies confirm the existence of the stabilizing ligands used in synthesizing the catalyst complex. These feature peaks are not present in the surface FTIR, indicating the loss of all organic ligands during electrodeposition process.

2.4.3 Surface morphology and structure characterizations of immobilized catalyst

As reported earlier, the FTIR results showed that the stabilizing ligands were lost during the deposition process. To determine if the Co-OXO WOC was successfully immobilized, we acquired SEM images at different magnifications, and EDS spectra of the Co-OXO-coated ITO samples. Figure 2.3 shows that the surface morphology of Co-OXO coated on ITO is uniform, although there is some degree of agglomeration. By contrast, the SEM image of bare ITO (Figure 2.3) at the same magnification shows a featureless surface. Figure 2.3c shows the EDS spectrum of the Co-OXO-coated ITO sample, which confirmed the existence of Co by the strong Co peaks. When the data provided by the SEM images and EDS spectra are taken together, we can confirm that the Co-OXO catalytic core was successfully immobilized on the ITO surface.



Figure 2.3. a) SEM image of the ITO surface following catalyst immobilization. It gives clear evidence of attachment of the catalyst to the surface. By contrast, the SEM image of ITO prior to catalyst deposition (Figure 2.3a inset) at an equivalent magnification has a featureless surface. b) SEM image of a selected area at a higher magnification. c) EDS diagram of Co-OXO-coated ITO sample, confirming that the catalyst core remained on the ITO substrate during the electrodeposition process.
2.4.4 Catalytic functionality, stability and recyclability assessments

Although we have confirmed that the catalytic core of Co-OXO WOC can be successfully immobilized via the electrodeposition method we developed, it was still necessary to determine whether the catalytic function of the catalyst has been affected by the loss of the stabilizing ligands. We did this by assessing and comparing the catalytic performance of both the heterogeneous and pseudo-homogeneous systems under identical conditions using cyclic voltammetry. As shown in Figure 2.4, we took CV scans of four systems: the catalyst coated on the ITO electrode (green), catalyst coated on gold (red), Co-OXO in a pseudo-homogeneous system (purple), and a bare ITO electrode (blue) to serve as baseline and control. The Co-OXOcoated ITO, Co-OXO-coated gold and pseudo-homogeneous systems have about the same onset water oxidation potentials, approximately 1.0 V vs. Ag/AgCl. We did a more careful examination of the data (Figure 2.4 inset), e to get more accurate estimates of onset potentials 0.93, 0.95 and 1.0 V, respectively, for pseudo-homogeneous, catalyst-coated ITO, and catalyst coated gold. As expected, bare ITO has a flat profile with no significant water oxidation functionality. Based on the analysis above, we can conclude that although Co-OXO WOC lost its stabilizing ligands during the electrodeposition process, its catalytic function remained intact.



Figure 2.4. Cyclic voltammograms acquired at a scan rate of 10 mV/s on bare ITO (blue curve), Co-OXO immobilized on gold (red), and Co-OXO immobilized on ITO (green) in 0.1M neutral phosphate buffer solution at pH 7.0. This diagram shows that water oxidation occurs at approximately 1 V for Co-OXO on both gold and ITO. For comparison, we have also included the CV of the catalyst in a pseudo-homogeneous system (purple). The profiles of the three scans are essentially identical, providing strong evidence that Co-OXO maintains its catalytic function whether in solution, or immobilized on gold or ITO.

To test for catalytic stability, we acquired 14 consecutive CV scans on one catalyst sample (Figure 2.5). The peak current densities of all 14 CV scans fall in a narrow range (1.54±0.04 mA/cm²). The profiles are also nearly identical for all scans. The total duration of the 14 consecutive scans was 74 min. These facts provide strong evidence that the immobilized catalyst is stable and has reproducible performance over a relatively long period of time.

To test the reusability and recyclability of the catalyst, we obtained 8 CV scans using a single immobilized catalyst sample. As opposed to the stability experiment we described above, this sample was pulled out of the testing solution after each scan, cleaned with DI water and dried in air before being returned to the solution for the next test. The results are shown in Figure 2.6. The profiles lie in a narrow range $(1.47 \pm 0.05 \text{ mA/cm}^2)$, providing strong evidence that the immobilized catalyst has stable and reproducible performance upon recycle and reuse.



Figure 2.5. Voltammograms acquired at a scan rate of 10 mV/s over 14 cycles of catalyst reuse. To acquire these data, the same catalyst was left in place for 14 consecutive scans. The data show negligible variation in both current density and peak currents, providing strong evidence that the immobilized catalyst is very stable, with a reproducible performance.



Figure 2.6. Assessment of catalyst recyclability and reusability. In this test, we acquired eight CV scans on a single sample of Co-OXO catalyst coated on ITO. After each scan, the sample was taken out of the solution, rinsed with DI water, dried in air and reinserted into the solution for the next scan. As the data show, all the voltammograms are distributed over a small range of values, confirming a high degree of catalyst reusability, as well as reproducibility in performance.

2.5 Conclusions

In this study, we developed an electrodeposition approach that can successfully immobilize an OEC mimic Co-OXO WOC onto conductive surfaces to promote catalyst recovery and reusability. The results show that, although all the stabilizing ligands of the catalytic complex were lost during the electrodeposition process, the functionality of the catalyst remained intact. This study also shows that the immobilized catalyst has the same performance as one dissolved in solution. We believe that the separation of catalyst synthesis and immobilization into separate processes has the potential of enabling systematic design and optimization of catalysts. The Co-OXO used in this study is only one example of a category of cubane-like WOCs which can be synthesized *ex situ* and immobilized on different conductive surfaces via electrodeposition. This study has provided a reliable pathway for effective recycling of this category of catalysts.

CHAPTER 3. SYNTHESIS OF A MnO_X CATALYTIC FILM ON FTO BY DUAL-SERIES CYCLIC VOLTAMMETRY

3.1 Abstract

We have developed a new method to synthesize a manganese-based water oxidation catalyst in situ in aqueous media. This new method involved multiple cyclic voltammetry (CV) series, followed by programed calcination to increase crystallinity. The method provides powerful tunability of the electrodeposition parameters, potentially enabling independent control of the surface morphology and the oxidation state of the catalytic film. Our results show that the first CV series produces a film with a nanoscale surface morphology, while the following CV series improves catalytic performance by elevating the oxidation state of the film while largely maintaining the existing surface morphology. Scanning electron microscopy (SEM) and transmission electron microscopy (TEM) were used to characterize the morphology of the MnO_x film, cyclic voltammetry (CV) to assess catalyst functionality, and long term amperometry to assess the stability.

3.2 Introduction

As we described in the previous chapters, the water oxidation reaction is the rate-limiting step in the artificial photosynthesis (AP) process, and the major source of overpotential^{1,32,33}. Thus, an effective AP process requires highly effective water oxidation catalysts (WOCs)³⁴. In the past decade, various catalytic complexes capable of water oxidation have been synthesized. Most of these complexes were based on Ru³⁵⁻³⁷, Pd³⁸ or Ir³⁹⁻⁴¹, all of which are earth-rare. These heavy transition metal-based complexes are expensive and usually toxic, which greatly limits their use on a large scale. As a result, using light first row transition metals to replace them has become a

research focus. The potential choices of transition metals are cobalt (Co) and manganese (Mn). Many Co-based WOCs have been successfully developed and tested in the past decade, the most successful example being the 'Co-Pi' catalyst developed by Nocera and coworkers⁴². Other examples include the Co-OXO catalytic complexes we described in Chapter 2⁴³. In the last five years, manganese has gained popularity as a new base metal for WOCs because it can form more than 30 minerals, with most of them having water oxidation properties⁴⁴⁻⁴⁶. Also, manganesebased materials are usually very flexible and can be synthesized *in situ*^{26,28,47,48} or *ex situ*⁴⁹⁻⁵¹. More importantly, manganese is the key element in the reaction center of the natural oxygen evolution center (OEC) in plants.

Various methods have been reported for synthesizing manganese-based WOCs, both *in situ* and *ex situ*. *In situ* synthesis, especially by electrodeposition, usually benefits from a simple and easily-controllable setup. Also, the resulting catalysts usually enable much better recyclability. Typically, electrodeposition is carried out by using chronoamperometry^{52,53} or a single-range of potentials^{48,54}.

In this study, we have developed a new method to synthesize a manganese-based water oxidation catalyst *in situ* in aqueous media. This new method involves two series of cyclic voltammetry (CV), followed by calcination to increase crystallinity. This method provides great tunability of the electrodeposition parameters, enabling independent control of the surface morphology and the oxidation state of the catalytic film. The goal of the study was to achieve an enhanced oxidation state and a nanoscale surface morphology⁵⁵.

3.3 Materials and methods

3.3.1 Materials

Manganese acetate tetrahydrate (Mn(CH₃COO)₂ 4H₂O) was purchased from Sigma-Aldrich LLC (St. Louis, MO). Sodium phosphate monobasic (NaH₂PO₄) and sodium phosphate dibasic (Na₂HPO₄) were purchased from J.T. Baker (Phillipsburg, NJ). All chemicals were used without further modification. Fluorine-doped tin oxide (FTO)-coated glass (surface resistance ~7 Ω /cm²) was purchased from the Sigma-Aldrich Company LLC (St. Louis, MO). Prior to use, the FTO glass was cut into 2cm × 1cm rectangular chips, cleaned with acetone, and rinsed in DI water.

3.3.2 Electrodeposition of MnO_x on FTO

The electrodeposition process was performed using a typical three-electrode experimental setup on a CHI 660D electrochemical workstation (CH instruments Inc., Austin, Texas). During the deposition process, a blank FTO chip was used as the working electrode, indium tin oxide (ITO) as a counter electrode, and a Ag/AgCl as a reference electrode. The electrolyte was 25 mL of a 50 mM manganese acetate solution. The deposition process was done by conducting two consecutive CV series over different potential ranges. The first series was ten cycles of potential ramps ranging from 0 to 0.6 V, at a scan rate of 10 mV/s. The second CV series was five cycles ranging from 0.6 to 2.0 V at an identical scan rate. During the first CV series, a light yellowish film could be observed forming on the FTO surface. The film turned dark brown during the second CV series. After deposition, the chip was rinsed in DI water and dried in air. It was then calcined at 450°C for five hours following a temperature ramp at a rate of 10°C/min in a Lindberg/Blue M tube furnace (Thermo Fisher Science Inc., Waltham, MA), and then cooled to 25°C.

3.3.3 Surface morphology characterization

The surface morphology of the samples was characterized on a JSM 6610 LV SEM (JEOL Ltd, Tokyo, Japan) as well as a JSM 2500FM SEM from the same manufacturer. Tunneling electron microscopy (TEM) images were taken on a JEM 2200FS TEM (JEOL Ltd, Tokyo, Japan). X-ray diffraction (XRD) spectra were acquired on a Bruker Davinci D8 Advanced X-ray diffractometer (Bruker Corporation, Madison, WI).

3.3.4 Catalyst functionality assessments and comparisons

The functionalities of the synthesized catalytic films were assessed by cyclic voltammetry on the CHI 660D electrochemical workstation. In each test, the catalytic film was used as a working electrode, a platinum mesh as counter electrode and a Ag/AgCl electrode as reference. The electrolyte used for all the tests was 25 ml of a 0.1M neutral phosphate buffer solution (pH 7.0) to simulate natural water. Potential ranges for all the CV tests were between -0.2V to 1.4V, at a scan rate of 10 mV/s. In a control experiment, a bare FTO chip was also tested under identical conditions.

3.3.5 Catalyst stability assessment

The stability of the catalytic film was assessed by amperometry at a constant potential of 1.2 V for a period of six hours under identical conditions as described in the functionality test. A control experiment was carried out on a bare FTO chip.

Another stability test was carried out by performing 18 consecutive CV scans on a single catalytic film. The testing and CV conditions were identical to those used for the functionality tests described above.

3.4 Results and discussions

3.4.1 Surface morphology characterizations

Figure 3.1 shows the SEM images of the surface morphology of: a) bare FTO, b) MnO_x film synthesized by dual series CV (S0), c) MnO_x film synthesized by single series CV 0-0.6V (S1), and d) MnO_x film synthesized by single series CV 0-2.0V (S2), respectively. The bare FTO has a featureless but rough surface morphology. We believe that the surface roughness plays an important role in the formation of the MnO_x films, as discussed in the latter part of this chapter. Figure 3.1 b and the inset show that the morphology of the MnO_x film synthesized by dual series CV has a network of long intermingled nanofibers. The diameter of the fibers was estimated to be 10~20 nm. Figure 3.1 c shows the morphology of samples S1 and S2, respectively. S1 has a nanofibrous morphology, which is quite similar to sample S0. This similarity provides important evidence that the morphology of the catalytic film synthesized by the dual-series method was primarily developed during the first CV series, but remained largely intact during the second CV

series. On the contrary, the S2 catalytic film developed into a morphology of micron-scale clusters, which is not optimal for mass transfer. Our analysis provides preliminary but important evidence that the surface morphology could be controlled by tuning the electrodeposition parameters. Since the dual- or multi-series deposition protocol provides more controllable parameters than the traditional single-series protocol, it could become an invaluable tool to enable predictable tuning of surface morphologies that promote catalytic performance.



Figure 3.1. SEM images of the surface morphologies of the FTO substrate, and of MnO_x after different in-situ synthesis methods. Image a: Bare FTO, showing a significant degree of surface roughness. Image b: MnO_x after two series of electrodeposition by CV, followed by calcination; the surface is a network of interconnected fibers in both the SEM and TEM (inset) images. Images c and d: MnO_x after electrodeposition by single CV series (0-0.6 V and 0-2.0 V, respectively); image c has a similar morphology to image b, although a few small spherical clusters are visible; image d shows a network of larger micron-sized fibers with random orientations.

3.4.2 Catalytic functionality assessments

As described above, the catalytic functionalities of the three samples were assessed by CV in a 0.1 M neutral phosphate buffer solution (pH 7.0). The S0 sample has an onset potential for water oxidation of approximately 1.2 V vs. SHE (Figure 3.2). We should note that the current density of S0 exceeds 1 mA/cm² at 1.45 V vs. SHE, which is the typical standard required to rank a WOC as highly active ⁵⁴. For comparison, both the S1 and S2 samples were assessed under identical conditions. The S1 sample, which was synthesized by the single-series (0-0.6 V) method, produced the desired nanoscale surface morphology similar to the dual-series sample. However, the catalytic performance was much lower. We believe this was due to inadequate elevation of oxidation states, as discussed in the next section. The S2 sample has a better catalytic performance than S1, which we believe is due to the more enhanced oxidation state. However, the surface morphology of S2 consists of much larger micrometer-scale fibers that likely retarded electron transfer, leading to a poorer catalytic performance.

To further assess if the dual-series method produces a catalyst that is superior to the catalytic films synthesized by single-series protocols, we evaluated and compared three additional samples synthesized by single-series over different potential ranges: 0-0.4 V, 0-0.8 V, and 0-1.5 V (Figure 3.3). All samples were assessed under the same conditions described above. As shown in Figure 3.3, all the single-series samples have similar catalytic performances except the 0-0.4 V sample, which is essentially non-catalytic. We believe the inadequate development of oxidation state led to the poor performance of the 0-0.4 V sample. The catalytic performance of the dual-series sample is clearly superior in comparison to all the single-series samples. As expected, the FTO chip had no measureable water oxidation effects in the potential range tested.



Figure 3.2. Catalytic functionality of the three MnO_x films on FTO. The blue curve is the cyclic voltammogram of the MnO_x film synthesized by the dual-series CV method; the purple and red profiles are the single-series synthesis from 0-0.6 V and 0-2.0 V, respectively; and the green curve is for bare FTO under identical assessment conditions. The MnO_x catalyst synthesized by dual-series CV is clearly superior to the single series species, because it benefits from the nanostructure developed in the first series, and the elevated oxidation state acquired during the second series. The importance of nanostructure is further reflected by the fact that the 0-0.6 V catalyst performs slightly better than the 0-2.0 V sample, even though the latter is at a higher oxidation state.



Figure 3.3. Comparison of the catalytic performances of dual series and single session samples by cyclic voltammetry in 0.1M phosphate buffer (pH 7.0). All the single series samples were synthesized under identical conditions as described in the experimental session, with the only difference being the ranges of potential ramps. Bare FTO was used again as a control. All single-series samples had similar performances, except the 0-0.4V sample which has a much lower activity. The dual-series sample has a much better performance than all single-series samples, which we believe to be the result of both an elevated oxidation state and nanoscale surface morphology.

3.4.3 Proposed formation mechanism

As we discussed before, the catalytic film synthesized by the dual-series method benefits from both nanoscale surface morphology and enhanced oxidation state. The data suggest that the surface morphology of the catalytic film was mainly formed during the first CV series, and the oxidation state of the film was elevated during the second CV series.

Currently, we do not know the exact mechanism for the formation of this unique surface morphology, and can only propose a possible hypothesis. We speculate that the protrusions on the rough FTO surface serve as nucleation sites for the nanofibers. During the first CV series, the Mn²⁺ ions would first be oxidized to Mn(III) hydrolysis species which form small nuclei on the protrusions ⁵⁶. The small nuclei then grow and extend across the whole surface. During the second CV series, the existing hydrolysis species would be partly oxidized to higher Mn(IV) species while the surface morphology remains largely intact. A schematic of the proposed mechanism is shown in Figure 3.4.

To evaluate this possible mechanism, we took photographs and SEM images after selected stages of deposition. We observed a yellowish film forming on the FTO surface during the first CV series (Figure 3.5). The film then turned to dark brown during the second CV series. Figure 3.6 shows a series of SEM images at different magnifications of the samples shown in the previous figure. Figure 3.6 a shows the sample after two cycles of the first series. Small clusters are clearly observable on the protrusions on the FTO surface. The small clusters gradually developed into a complete film across the conductive surface (Figure 3.6 b-d). During the second CV series, the surface morphology of the catalytic film did not change much (Figure 3.6 e).



Figure 3.4. Cartoon illustration of the proposed formation mechanism. We propose that Mn^{2+} ions are first oxidized on the protrusions of the FTO surface during the first CV series, forming fibrous MnO(OH) species. During the second CV series, the oxidation state of the catalytic film is further elevated while the surface morphology remains largely intact.



Figure 3.5. Time-lapse photographs of MnO_x samples at selected stages of deposition. During the first CV series, a light yellowish film could be observed on the FTO surface. The film turned to dark brown during the second CV series.



Figure 3.6 a. SEM images at various magnifications of the sample after two (2) cycles in the first CV series. Small clusters can be observed on the protrusions on the FTO surface.



Figure 3.6 b. SEM images at various magnifications of the sample after four (4) cycles of the first CV series. The small clusters in Figure 3.6a have developed into larger clusters.



Figure 3.6 c. SEM images at various magnifications of the sample after six (6) cycles of the first CV series. The clusters in Figure 3.6 b continue to grow as the catalytic film is forming.



Figure 3.6 d. SEM images at various magnifications of the sample after eight (8) cycles in the first CV series. The catalytic film in Figure 3.6 c has now covered the whole surface.



Figure 3.6 e. SEM images at various magnifications of the sample after four (4) cycles in the second CV series. The surface morphology of the catalytic film did not change much during the second CV series.

We also did functionality tests after selected stages of deposition. As shown in Figure 3.7, there is a gradual increase in peak current density during the first CV series. However, the current density increased dramatically during the second CV series, which we believe is due to the increase in the oxidation state of the MnO_x film. It is particularly worth noting the slope of the functionality curve during the second session. There is clearly the possibility of a much further enhanced oxidation state with additional scans over this potential range, or with the use of a third series of electrodeposition over a different potential range.



Figure 3.7. Functionality assessment after selected stages of electrodeposition. There is a gradual increase in peak current density during the first CV series. However, the peak current density increases dramatically during the second CV series, which we speculate to be the result of continued enhancement of the oxidation state of the catalyst. It is particularly worth noting that the slope of the functionality curve during the second session is strongly positive, and nowhere near an asymptote. Thus, there is clearly a possibility of much further enhanced oxidation state with additional scans over this potential range (0.6 to 2.0 V) or a third series of CV electrodeposition over a different potential range, with the potential that this method can be used to optimize catalyst function.

3.4.4 XRD characterizations

Our XRD results provide important evidence for this proposition. The black, green and red traces in Figure 3.8 show the XRD patterns for samples S0, S1 and S2, respectively. The major components for the dual-series sample were Mn_2O_3 and MnO_2 , which is similar to sample S2. Sample S1, on the contrary, only has feature peaks for Mn_2O_3 . We speculate that the major transformation in the first CV series (0-0.6V) was the conversion of Mn(II) to Mn(III), with the conversion of Mn(III) to Mn(IV) taking place during the second CV series (0.6-2.0V). Since there was only a very small number of cycles (5 cycles) during the second series, some Mn(III)species remained intact. We should also note that the XRD profile of sample S2 has the same oxidation state as the dual-series catalyst, so its poorer performance could be attributed entirely to transport resistances due to the larger microscale morphology.



Figure 3.8. X-ray diffraction spectra of the three catalysts: a) MnO_x after two consecutive series of CV electrodeposition (scans over 0-0.6 V and 0.6-2.0 V) followed by calcination; it has the peaks of both Mn_2O_3 and MnO_2 , which indicates that the MnO(OH) formed in the first CV series has been further oxidized. b) MnO_x after one series of CV (0-2.0 V) and calcination, showing a similar pattern to trace a); this is an indication that the single 0-2.0 V series produces a MnO_x catalyst with a similar oxidation state as the one synthesized by the dual-series. c) MnO_x after one CV series (0-0.6 V) and calcination; the scan shows that the major phase of MnO_x is Mn_2O_3 , meaning that the MnO_x before calcination is MnO(OH) as proposed. The symbols +, *, and # in this figure designate the feature peaks of the FTO substrate, Mn_2O_3 and MnO_2 , respectively.

3.4.5 Stability tests

Long and stable catalytic performance is also a very important aspect of functional WOCs ⁵⁷. The catalytic film synthesized by dual-series methods has very stable catalytic performance during consecutive CV tests and chronoamperometry tests over prolonged periods of time. All the tests were performed using the identical setup and conditions described in the functionality assessments session. Figure 3.9 shows the current densities when 1.2 V (vs. SHE) of potential was applied to the system. After about 1.5 h of decrease in signal (which is typical⁵⁸), the current density was steady for the next 4.5 hours of the six-hour assessment period. As expected, FTO has a flat profile throughout the assessment period, confirming that it is not catalytic.

We performed the same test on samples S1 and S2. The results of these tests show that they are equally stable but at lower current densities than the dual-series sample (Figure 3.10). We also did stability tests with 18 consecutive CV cycles on one S0 sample. Figure 3.11 shows that all the resulting curves were of similar shape, with the peak current densities falling within a narrow range $(1.96\pm0.11 \text{ mA/cm}^2)$.



Figure 3.9. Current density profile of the MnO_x catalyst synthesized by the dual-series method during long term catalytic assessment at 1.2 V (vs. SHE) in a 0.1 M phosphate buffer solution (pH 7.0). The steady profile over a long period of time gives an indication of a stable MnO_x film.



Figure 3.10. Current density profiles showing the stability of the catalytic films: bare FTO (purple curve), single session 0-0.6V sample (black), single session 0-2.0V sample (red), and dual session sample (blue). All tests were carried out in a 0.1M phosphate buffer solution (pH 7.0) at 1.2V (vs. SHE) for six hours. All single session and dual session samples had good stability during the six hours of assessment. However, the dual session sample had a higher stable current density compared to the single session samples, providing further evidence that the dual series method is better than the traditional single session methods.



Figure 3.11. Consecutive CV scans on one catalyst sample to assess stability. All 18 scans have similar profiles, with the peak current density falling within a narrow range of values $(1.96\pm0.11 \text{ mA/cm}^2)$

3.5 Conclusions

We have developed and assessed a new electrodeposition method for synthesizing stable and highly efficient manganese-based WOCs *in situ* on conductive surfaces. This new method involves multiple consecutive series of CV scans over different potential ranges, followed by calcination. The resulting MnO_x films have well-dispersed porous surface morphologies consisting of nanoscale fibers that promote catalytic performance. Our characterization results support a hypothesis that the first CV series produces a fibrous nanoscale film, while the second series elevates the oxidation state of the catalyst without significantly changing the surface morphology acquired during the first series. The potential to enable independent control of surface morphology and catalyst oxidation state(s) makes this a powerful tunable method for rational catalyst synthesis and optimization.

CHAPTER 4. FUNCTIONAL BIOMIMETIC COMPOSITE Mn/TiO₂ PHOTOANODE FOR WATER SPLITTING

4.1 Abstract

We have designed, fabricated and assessed a composite photoanode for solar-driven water splitting, using an easy approach and inexpensive materials. The photoanode consists of a solar-harvesting layer and a catalytic layer. Scanning electron microscopy (SEM) and x-ray powder diffraction (XRD) were used to characterize the surface morphology and structure, and cyclic voltammetry (CV) was used to assess functionality with and without ultraviolet (UV) illumination. The resulting photoanode provides stable and effective water oxidation under UV illumination. Another potential benefit of this work is that, since TiO₂ is used as the solar-harvesting layer, various dyes can be integrated into the device to broaden its absorption spectrum.

4.2 Introduction

In the previous chapters, we discussed work on developing cost-effective, highly active water oxidation catalysts. In this chapter, we focus on designing, fabricating and assessing a composite photoanode for solar-driven water splitting, using a simple approach and inexpensive materials.

As we described in Chapter 1, a functional artificial photosynthesis device requires an effective solar-driven water oxidation photoanode. A functional photoanode should incorporate the following features: i) a robust and conductive substrate upon which components like a solar harvester and a WOC can both be mounted; ii) a solar harvester that can effectively capture, intensify and convert sunlight from a target range of the solar spectrum; iii) an effective WOC

that is able to effectively catalyze the water oxidation reaction. A double-layer structure is the best configuration for the photoanode, because it has the potential to enable independent design and optimization of the individual components, which could enhance the overall performance of the integrated system.

We had initially planned to integrate a dye-sensitized solar cell (DSSC) with a manganese-based WOC to build the photoanode system. DSSC is a flexible solar converter system that has tunable absorption spectra when different types of dyes are incorporated ⁵⁹⁻⁶¹. A DSSC usually has a TiO₂ support as a substrate incorporated with dyes ⁶²⁻⁶⁵. TiO₂ by itself has a large bandgap (~ 3eV), which is sufficient for the water oxidation reaction ⁶⁶⁻⁶⁸. In fact, Fujishima and Honda used TiO₂ under intense ultraviolet (UV) light illumination in their pioneering work on using photoelectrochemistry to split water ⁶⁹. However, as shown by Equation 4.1, this bandgap limits its absorption spectrum to no more than 413 nm, which falls in the ultra-violet region.

$$E = \frac{hc}{\lambda}$$

$$\lambda = \frac{hc}{E}$$

$$\lambda = \frac{6.63 \times 10^{-34} J \cdot s \times 3 \times 10^8 m/s}{3 \times 1.60 \times 10^{-19} J} = 4.14 \times 10^{-7} m = 414 nm \quad (Eq 4.1)$$

On the other hand, TiO_2 has a low and negative conductive band, which is unlike many other semiconducting materials, and can thus enable water splitting without an external electrical bias

 70,71 . TiO₂ also provides excellent mechanical, thermal and chemical stability under the harsh conditions typically encountered in water oxidation. Finally, TiO₂ can be easily engineered into various morphologies and coated onto different surfaces via a variety of methods. All these benefits make TiO₂ an ideal substrate for the solar harvesting system. To overcome the limitation of the limited absorption spectrum, various dyes can be integrated with TiO₂ to tune and broaden the spectrum and boost the efficiency of the solar harvesting system.

Our goal was to demonstrate the feasibility of building a photoanode using a manganese-based WOC and TiO_2 as a solar harvesting material. If the composite photoanode is capable of solardriven water splitting, then we can simply add various dyes to improve the absorption spectrum and thus the performance.

We took advantage of the desirable properties of both TiO_2 and manganese-based WOC to develop the photoanode. We then performed electrochemical measurements to assess the functionality of the photoanode, and used SEM and XRD to characterize the structure and surface morphology.

4.3 Materials and methods

4.3.1 Materials

Manganese (II) acetate tetrahydrate ($Mn(CH_3COO)_2 \bullet 4H_2O$) and P25 titanium dioxide nanoparticles (TiO₂ with an anatase to rutile ratio of 4:1) were purchased from the Sigma-Aldrich Company, LLC (St. Louis, MO) and used without further modification. Fluorine-doped tin oxide (FTO with a surface resistance of ~ 7 Ω /cm²) was also purchased from Sigma-Aldrich. Prior to use, the FTO glass was cut into 2cm by 1cm rectangular substrates, and cleaned in acetone and DI water.

4.3.2 Methods

4.2.2.1 Spin coating of the TiO₂ film

We added 0.15g of P25 TiO₂ nanoparticles to 10 ml of methanol. The mixture was then sonicated for 10 minutes to make a P25 colloid. The colloid was then spincoated onto FTO glass for 20 minutes. The TiO₂-coated FTO glass was then dried in air.

4.3.2.2 In situ synthesis of the Mn WOC catalyst

The Mn WOC was synthesized by the electrodeposition protocol reported by Zaharieva and coworkers (2012) ⁵⁴. The electrodeposition was performed using a typical three-electrode setup on a CHI 660D electrochemical workstation (CH Instruments, Inc., Austin, TX). During the deposition, TiO₂-coated FTO was used as the working electrode, an indium tin oxide-coated polyethylene (ITO) chip as the counter electrode, and a Ag/AgCl electrode as the reference electrode. Twenty five milliliters (25 ml) of a 0.1 M Mn(CH₃COO)₂ aqueous solution was used as electrolyte. For electrodeposition, 25 cycles of potential ramps ranging from -0.75 V to +2.15 V were applied at a scan rate of 0.1 V/s. A light brown film could be observed forming on the electrode surface, with the film becoming darker during the deposition. After deposition, the photoanode sample was rinsed with D.I. water and dried in air. For comparison, we also

synthesized a Mn WOC on a bare FTO electrode without the TiO₂ coating, using the identical approach described above.

4.3.2.3 Structural characterization by XRD and SEM

We took XRD spectra after each deposition step to track the color changes on the surface. The XRD spectra were acquired on a Bruker Davinci D8 advanced x-ray diffractometer (Bruker Corporation, Madison, WI). We also used a JSM 6610-LV scanning electron microscope (JEOL Ltd, Peabody, MA) to observe the surface morphology, and EDS (Oxford Instruments, High Wycomb, Bucks, England) to determine the surface elemental composition.

4.3.2.4 Functionality assessment by CV and EIS

We did CV and EIS characterizations using a typical three-electrode electrochemical setup. In all the assessments, a platinum wire was used as a counter electrode, a Ag/AgCl electrode as reference electrode and 25ml of 0.1 M neutral phosphate solution (pH 7.0) as electrolyte. We applied a potential range of -0.2 V to 1.4 V at a scan rate of 10 mV/s.

We first tested the Mn WOC that was synthesized directly on clean FTO without TiO₂ coatings, to make sure that the electrodeposition process we described above can produce an active catalyst. As comparison, we used a piece of clean FTO glass as control. We then tested the composite photoanode samples in the dark and under UV illumination. The UV source was a 4 W UVGL-15 compact UV lamp (UVP LLC, Cambridge, UK). For the EIS tests, we collected the impedance spectrum on a photoanode sample under UV illumination at 0.8 V, using a 10 mV perturbation between 0.1 Hz and 10 kHz.

4.4 Results and discussion

4.4.1 Structural and morphological characterization of the composite photoanode

To track changes on the FTO surface during the photoanode fabrication process, we took XRD spectra after each step. Shown in Figure 4.1 are the XRD patterns for a) pure P25 TiO₂ powder, b) pure FTO substrate, c) FTO substrate coated with TiO₂, and d) the composite photoanode with TiO₂ coating and Mn WOC. Trace 'a' is a typical pattern of TiO₂ ⁷², and trace 'b' a typical pattern of the FTO substrate. When we compare traces 'a', 'b', and 'c,' we can conclude that TiO₂ was successfully immobilized on FTO glass using the method described above. We expected to see the feature peaks for manganese in trace 'd' to differentiate it from trace 'c.' However, there were no significant differences between the two profiles. There are two plausible explanations for this result: i) it is difficult for XRD to detect the manganese-based catalyst on the surface; and ii) the *in situ* synthesis did not deposit the manganese catalyst as had been expected.



Figure 4.1. XRD spectra of a) pure P25 TiO₂ powder, b) pure FTO glass, c) FTO glass coated with TiO₂, and d) the composite photoanode. In this figure, + designates the feature peaks of TiO₂, and * denotes those of FTO. Traces "a" and "b" show the typical patterns for TiO₂ and FTO, respectively. Trace "c" shows all the peaks associated with both TiO₂ and FTO, confirming that the composite photoanode contains all the required elements. However, trace "d" is not significantly different from trace "c," which could mean that the method described above did not produce the expected composite catalyst, or that XRD is not an appropriate technique for detection of manganese in the complex.
To resolve this issue, we took EDS images of the photoanode to check the structure of the surface. Figure 4.2 shows SEM images of: a) clean FTO surface, b) FTO coated with TiO₂ at low (x200) magnification, c) FTO coated with TiO₂ at a higher (x800) magnification, d) the composite photoanode at low (x160) magnification, and e) the composite photoanode at a higher (x3300) magnification. Compared with bare FTO, which has a flat featureless surface, image b shows the surface has been covered by a film of TiO_2 . However, instead of a uniformly smooth coated layer, we observed many cracks. We speculate that these cracks were formed during the evaporation of the solvent, likely similar to the mechanism for the formation of cracks on the surface of the ground during a long drought. This morphology can be seen more clearly in image c. Image d shows the surface morphology after electrodeposition of the WOC. The morphology appears to be similar to the previous samples with only TiO₂ coatings. However, at a higher magnification (3e), we observed a different coating layer of distinct morphology beneath the TiO₂ substrate. We believe this layer beneath the TiO₂ coating is the manganese WOC film. Since this film was covered by the TiO₂ coating layer, it could not be effectively detected by XRD, which is primarily a surface characterization technique. To further investigate this, we took EDS elemental mapping of a selected area where both layers could be clearly observed.



a b c $\int_{U_{u_{u_{u_{u}}}}} \int_{U_{u_{u}}} \int_{U_{u_{u}}} \int_{U_{u_{u}}} \int_{U_{u_{u}}} \int_{U_{u}} \int_{U_{u}}$

Figure 4.2. SEM images of a) clean FTO substrate; b) FTO surface after coating with TiO_2 (at x200 magnification); c) FTO coated with TiO_2 (at x800 magnification); d) composite photoanode (at x160 magnification); e) composite photoanode (at x3,300 magnification). Images b and c show the morphology of the TiO₂ coating to be essentially a uniform layer, with a porous structure that contains cracks. Images d and e show the morphology of the composite photoanode. Following Mn WOC deposition, the morphology of the surface does not change substantially; however, a new layer could be observed underneath the porous TiO_2 film.

Figure 4.3 is the SEM image of the area we selected for EDS examination. The triangular area in this image is the top TiO_2 coating layer, while the rest of the area is the 'crack' in the TiO_2 layer that shows the film underneath. The elemental distribution of titanium and manganese are shown in Figure 4.3 b and c, respectively. Figure 4.3 b shows a high concentration of titanium in the triangular area, with very little manganese detectable. On the other hand, the white dots in Figure 4.3 c clearly portray the presence of manganese in the area that is not covered by TiO_2 . This is an interesting observation because, since TiO_2 was coated prior to the electrodeposition of the Mn WOC, we should expect the Mn WOC to be deposited above the TiO_2 layer instead of below. We believe this anomaly occurred because the Mn(II) ions migrated through the porous TiO_2 layer underneath the layer of TiO_2 coating.



Figure 4.3. EDS mapping of selected area. a) The SEM image provides evidence of both TiO_2 and manganese on the surface. The triangular area at the top is presumed to be the TiO_2 layer, while the rest of the area is covered by the manganese catalyst. b) The distribution of titanium within the selected area. As postulated, the highest accumulation of titanium is in the triangular area. c) Distribution of manganese in the selected area, with the heaviest concentration outside the triangular area.

4.4.2 Electrochemical characterization of the Mn WOC

According to a previous report,⁵⁴ the catalyst comes in active and inactive forms. To determine which form is generated by the method described above, we acquired CV data on pure FTO, and FTO samples coated with Mn WOC. The blue curve in Figure 4.4 represents the CV for the Mn WOC, and the red curve the bare FTO substrate. These data lead to the conclusion that the Mn WOC synthesized using this protocol is active, with an onset potential for water oxidation of approximately 1.1 V vs. Ag/AgCl. We have tested the Mn WOC sample both with and without UV illumination. Similar results were obtained under both conditions, confirming that the Mn WOC is not light-sensitive.



Figure 4.4. Typical cyclic voltammograms during assessment of the manganese catalyst. The red trace represents data for clean FTO glass. The blue trace represents data acquired on the manganese catalyst deposited on FTO, and provides confirmation that the catalyst synthesized by electrochemical deposition is active.

4.4.3 Electrochemical characterization of the composite photoanode.

As described above, TiO_2 has a band gap of ~3 ev which corresponds to an absorption spectrum in the UV range⁷³. Therefore, we characterized the photoanode with and without UV illumination. To help identify the inflection point in the curve, we took the first derivative of the current density with respect to the applied potential (Figure 4.5). The point where the first derivative goes to zero can be used as a reliable estimate of the onset potential for the water oxidation reaction. Figure 4.5 a shows the profile of the photoanode sample under UV illumination, while Figure 4.5 b shows the sample without UV illumination. Without UV illumination, the onset potential was estimated at approximately 1.1 V vs. Ag/AgCl, demonstrating that the functionality of the photoanode is similar to that of the pure Mn WOC. However, under UV illumination, the onset potential is reduced by 0.3 V to about 0.8 V vs. Ag/AgCl. This decrease of onset potential represents enhancement of the catalytic functionality under radiation. We also tested the sample with only the TiO₂ coating under identical conditions but, as expected, we observed no measurable catalytic activity. As mentioned before, the Mn WOC by itself is not light-sensitive, thus the decrease of the onset potential should only be the result of the TiO₂ layer.



Figure 4.5. First derivative of the current density profile with respect to applied potential, used here as a reliable aid to determining the oxidation voltage. a) Data acquired under UV illumination, showing an oxidation voltage of approximately 0.8 V vs. Ag/AgCl. b) The oxidation voltage for the same sample evaluated without UV illuminationwas of approximately 1.1 V vs. Ag/AgCl.

As described above, the photoanode should have a "diffusion-reaction" mechanism, where the reactants diffuse through the porous TiO_2 layer to the surface of the catalytic film where they react, and the products diffuse in the opposite direction. To investigate this scheme, we acquired EIS data at 0.8V vs. Ag/AgCl under UV illumination. The resulting Nyquist plot is shown in Figure 4.6. The plot of the photoanode under this condition has a semicircle in the high frequency range, followed by a 45 degree straight line in the low frequency range. A simple Randles Circuit (Figure 4.5 inset) can be used to regenerate the Nyquist plot. The equivalent circuit contains a surface resistance R_{s} , a double-layer capacitance C_{dl} , a charge transfer resistance R_{ct} , and a Warburg element Z_w which represents the diffusion process. A Randles circuit is a typical model for an electrochemical reaction when it is governed by a semi-infinite diffusion process on a flat electrode. This result further confirms our hypothesis that, during the deposition of the Mn WOC, Mn^{2+} ions diffused through the TiO₂ layer, and formed the WOC catalytic film underneath. Thus, although the catalytic layer is covered by TiO₂, water molecules can still reach the catalytic surface to enable the reaction.



Figure 4.6. Nyquist plot of EIS data acquired at 0.8 V for the composite photoanode under UV illumination. The semicircle at high frequency denotes reaction, while the straight line at low frequency denotes diffusion. Along with the Randles circuit (inset) used to simulate the data, this result provides confirmation that the Mn WOC is covered by a porous layer of TiO_2 .

4.5 Conclusions

We have successfully developed a composite photoanode using a Mn WOC and TiO₂ as a conductive substrate. This work has demonstrated an important and practical approach to using earth-abundant, non-toxic, easy to handle catalysts to fabricate a composite photoanode. Another potential application of this work that is that this catalytic complex can be integrated with CO_2 sequestration to achieve the two carbon-based-fuel production schemes we introduced in Chapter 1. The current photoanode is only functional under UV illumination. However, there are methods to dope various types of substrates with selected dyes to build a DSSC-type solar harvester to broaden the absorption spectrum of the photoanode to obtain a much more efficient system.

CHAPTER 5. ELECTRODEPOSITION OF Ni/Ni(OH)₂ CATALYTIC FILMS BY CYCLIC VOLTAMMETRY FOR THE HYDROGEN EVOLUTION REACTION

5.1 Abstract

We have developed a novel electrodeposition method that synthesizes Ni/Ni(OH)₂ catalytic films *in situ* on conductive surfaces. The new method involves only two cycles of cyclic voltammetry, and the resulting film has a surface morphology consisting of a monolayer of firmly packed walnut-shaped particles. The film has superior catalytic activity towards the hydrogen evolution reaction compared to traditional hydrogen evolution reaction catalysts (HERCs) produced by electrodeposition under constant current or constant potential. The method is a simpler and more cost-effective way to produce highly-effective electrocatalysts for the hydrogen evolution reaction (HER). It creates large numbers of Ni(OH)₂/Ni interfaces that significantly promote catalyst performance towards that of platinum mesh.

5.2 Introduction

As we described in Chapter 1, active and low-cost electrocatalysts for both the water oxidation and hydrogen evolution reactions are crucial for an effective AP process to produce hydrogen by water splitting^{33,74-76}. For a long time, the water oxidation half-reaction attracted most of the attention because it is the major source of overpotential in the water splitting process⁷⁷. Our group has also worked extensively on the water oxidation reaction, as discussed in the previous chapters. For the hydrogen evolution half reaction, platinum and its alloys are considered the "state of art" catalysts, and are widely used at the laboratory scale⁷⁸⁻⁸⁰. However, the high cost and extremely low availability of platinum and its alloys render them impractical for use on a large scale^{81,82}. To make affordable artificial photosynthesis devices at scale, highly effective, robust catalysts based on earth-abundant metals have to be developed. Over the last decade, HER catalysts based on earth-abundant transition metals, especially nickel, have been extensively studied and assessed⁸³⁻⁹³. Most of these Ni-based HERCs are synthesized *ex situ* through bench top wet chemistry methods. As previously discussed, electrodeposition has the benefit of significantly simplified experimental setup as well as easily-controllable synthesis conditions. It can also promote catalyst recovery and recyclability, which are usually significant problems for traditional catalysts synthesized by benchtop methods.

In this chapter, we discuss the development of an electrodeposition method to produce Ni/Ni(OH)₂ catalytic films *in situ* for hydrogen evolution on conductive surfaces. The new method involves only two cycles of cyclic voltammetry over a voltage range of 0 V to -1.2 V vs. Ag/AgCl (immersed in 3M NaCl). We believe the method provides a simpler and more inexpensive way to fabricate highly-effective hydrogen evolution electrocatalysts.

5.3 Experimental session

5.3.1 Materials

Nickle acetate tetrahydrate (Ni(COOCH₃)₂ · 4H₂O) was purchased from the Sigma Aldrich Company, LLC (St. Louis, MO). Sodium phosphate monobasic (NaH₂PO₄) and sodium phosphate dibasic (Na₂HPO₄) were purchased from J.T. Baker (Phillipsburg, NJ). All chemicals were used without further modification. Fluorine-doped tin oxide (FTO)-coated glass (surface resistance ~ 7 Ω /sq) was also purchased from the Sigma-Aldrich Company, LLC (St. Louis, MO). Prior to use, the FTO electrode was cut into $2\text{cm} \times 1\text{cm}$ rectangular chips, cleaned in methanol and acetone, and rinsed in D.I. water.

5.3.2 Synthesis by electrodeposition

The electrodeposition process was performed using a typical three-electrode setup on a CHI 660D Electrochemical Workstation (CH Instruments Inc., Austin, Texas). A blank FTO chip was used as the working electrode, with an indium tin oxide (ITO) chip as a counter electrode, and a Ag/AgCl electrode as reference. The electrolyte used for electrodeposition was 25 ml of a 150 mM Ni(OCOCH₃)₂ solution. Two cycles of cyclic voltammetry (CV) scans were conducted between 0.0 V and -1.2 V vs. Ag/AgCl at a scan rate of 0.01 V/s. The cumulative charge was ~1.0 C. During the electrodeposition process, a light grayish film could be observed on the FTO working electrode. After deposition, the chip was rinsed in DI water and dried in air.

For comparison, we also synthesized a Ni-based HER catalyst using the traditional electrodeposition method of constant current density of 8 mA/cm² for 125 seconds^{94,95}. A shiny silvery film could be observed after the deposition. We also compared the performance of the Ni/Ni(OH)₂ catalytic film to that of platinum mesh. For brevity, we will designate the Ni/Ni(OH)₂ catalyst as cv-Ni, and the traditional catalyst as cc-Ni.

5.3.3 Materials characterization

The surface morphologies of both samples were characterized by a JSM 6610-LV and a JSM 7500F scanning electron microscope (SEM) (JEOL Ltd, Tokyo, Japan). The detailed

morphology was evaluated by a JEM 2200FS transmission electron microscope (JEOL Ltd, Tokyo, Japan). X-ray diffraction spectra were acquired on a Bruker Davinci D8 Advance X-ray diffractometer (Bruker Corporation, Madison, WI).

5.3.4 Electrochemical assessment of functionality

The functionality of the samples was assessed by cyclic voltammetry on a CHI660D electrochemical workstation. FTO glass coated with cc- or cv-Ni was used as a working electrode, a platinum wire as counter electrode, and a Ag/AgCl electrode as reference, with 25 ml of 1 M KOH solution as electrolyte. Potential scans were conducted between -0.4 V and -0.8 V vs. Ag/AgCl at a scan rate of 0.01 V/s. A Tafel plot was also obtained at a scan rate of 0.01 V/s. The catalytic functionalities of all samples were assessed in neutral media with 25 mL 0.1 M phosphate buffer (pH 7.0) as electrolyte. For comparison, a Ni foil and a platinum mesh were also assessed.

The catalyst stability test was also performed using the setup described above. The potential was maintained at -1.0 V for 10 hours and the corresponding current density was recorded as a function of time. We also did 201 cycles of CV scans on one cv-Ni sample, with the same setup used for the functionality tests in neutral media.

5.4 Results and discussions

5.4.1 Structural characterization

Figure 5.1 shows the surface morphology of the FTO substrate and the two samples obtained by FESEM. The bare FTO again has a rough but otherwise featureless surface (Figure 5.1 a). The cc-Ni sample also has a featureless surface (Figure 5.1 b). On the contrary, the cv-Ni sample has a surface morphology consisting of a monolayer of evenly distributed, firmly packed spherical walnut-shaped particles (Figure 5.1 c). The inset in Figure 5.1c shows the surface morphology of the cv-Ni sample at a higher magnification. From this image, we can estimate the diameter of the spherical particles at approximately $0.1 \sim 0.2 \mu m$ (100-200 nm). We believe the uniform surface morphology promotes mass transfer and enhances the catalytic performance of the cv-Ni catalyst. Based on recent reports, we believe the edges on the particles are most likely nano-walled Ni(OH)2^{96,97}.

To determine the elemental composition of both cc- and cv-Ni samples, energy dispersive x-ray spectra (EDS) were obtained on the two catalytic films. To eliminate the influence of oxygen from the underlying tin oxide substrate, we scratched off the catalytic films before EDS measurements. The EDS spectrum of cc-Ni only shows the signal for Ni (Figure 5.1 d). This indicates that the cc-Ni catalyst is a purely metallic nickel film. On the other hand, the cv-Ni sample (Figure 5.1 e) has signals for both Ni and O, which provides evidence that the spherical particles have a composite structure with both Ni and Ni(OH)₂ present. Figure 5.2 shows the cross-section SEM image of cv-Ni sample. The thickness of the catalyst layer was estimated at ~ 0.3μ m. It also confirms that the surface morphology of the cv-Ni sample is a monolayer of spherical particles.



Figure 5.1. SEM images and EDS characterization of cc-Ni and cv-Ni. a) SEM image of a bare FTO substrate, showing a featureless but rough surface. b) SEM image of the cc-Ni sample, showing a featureless metallic surface. c) cv-Ni at an identical magnification, showing a film consisting of a monolayer of evenly-distributed walnut-shaped particles with diameters of $0.1 \sim 0.2 \mu m$. The inset is the surface of the cv-Ni sample at a higher magnification, showing a detailed morphology of spherical particles with patterns on the surface. d) EDS characterization of cc-Ni, showing only the signal for Ni, which is an indication that it consists mainly of a Ni film. e) EDS characterization result for cv-Ni showing both Ni and O, an indication that cv-Ni contains both Ni and Ni(OH)₂. The carbon signals in both EDS spectra are from supporting SEM slabs and patches.



Figure 5.2. Cross-sectional SEM image of the cv-Ni sample. The thickness of the Ni/Ni(OH)₂ catalytic film was estimated at $\sim 0.3 \mu m$. This also confirms that the surface morphology of the catalyst layer is a monolayer of firmly packed spherical particles.

Figure 5.3 shows the XRD spectra of bare FTO and both Ni-based catalyst samples. In all spectra, + designates the feature peaks for the conductive substrate FTO, * the peaks for Ni, and # the peaks for Ni(OH)₂. As expected, both the cc- and cv- Ni samples clearly show the feature peaks for Ni. However, the cv-Ni sample also shows additional small peaks at 34, 38 and 44 degrees, which are indicators of the existence of $Ni(OH)_2^{98-100}$. Due to the relatively small amount of Ni(OH)₂ compared to much larger quantities of FTO and Ni, those peaks are not very prominent.



Figure 5.3. XRD spectra of a) cv-Ni, b) cc-Ni and FTO films. The symbol + designates feature peaks of FTO, * of Ni, and # of Ni(OH)₂. The cc-Ni film has peaks of FTO and Ni, while the cv-Ni spectrum has peaks of FTO, Ni and Ni(OH)₂. Due to the small amount of Ni(OH)₂ in the cv-Ni samples compared with FTO and Ni, the Ni(OH)₂ peaks are not prominent.

5.4.2 Functionality assessments

Cyclic voltammetry was used to assess and compare the functionalities of the cc- and cv-Ni samples, and Pt mesh in neutral and alkaline media. Figure 5.4 shows the CV spectra acquired in 0.1M neutral phosphate buffer. The black curve shows the current density of cc-Ni in response to applied potential. The red and green curves are for cv-Ni and platinum mesh, respectively. Although both the cc- and cv-Ni samples have higher onset potentials compared with Pt mesh, cv-Ni is much closer to platinum than cc-Ni. This result indicates that cv-Ni has a better catalytic performance than cc-Ni. The peak current density of cv-Ni (0.806 mA/cm²) is much larger than for cc-Ni (0.211 mA/cm²), based on geometric surface area at -0.3V vs. NHE. We also assessed the samples in 1 M KOH alkaline media (Figure 5.5). Again, the peak current density of cv-Ni is about twice that of cc-Ni, which demonstrates much better catalytic performance.



Figure 5.4. Comparison of the functionality of cv-Ni, cc-Ni, and platinum mesh by cyclic voltammetry in neutral media. The black, red, and green curves are for cc-Ni, cv-Ni, and Pt mesh, respectively. Both cc- and cv- Ni have higher onset potentials than Pt, but the cv-Ni catalyst has a lower onset potential than cc-Ni.



Figure 5.5. Comparison of the functionality of cv-N1, cc-N1, and platinum mesh by cyclic voltammetry in alkaline media. The black, red, and green curves are for cc-Ni, cv-Ni, and Pt mesh, respectively. Both cc- and cv- Ni have higher onset potentials compared to Pt, but the cv-Ni catalyst has a lower onset potential than cc-Ni.

We compared the Tafel slopes and exchange current densities of the cc- and cv-Ni catalysts in 0.1 M neutral phosphate buffer. The cc-Ni has a slope of 0.34 V/dec, while cv-Ni has a much lower slope of 0.25 V/dec (Table 5.1). As comparison, the Ni foil has a slope of 0.35 V/dec, which is very close to the value for cc-Ni. We believe this offers further evidence that cc-Ni is a simple metallic Ni film, while cv-Ni has a Ni/Ni(OH)₂ composite structure. The cv-Ni also has a higher exchange current density compared with cc-Ni, which is an indication of much better catalytic performance.

Table 5.1. Comparison of Tafel slopes and exchange current densities of cc-Ni, cv-Ni, and Ni foil catalysts assessed in neutral media. The lower slope and higher exchange current density of cv-Ni are indicators of much better catalytic performance.

Sample	Slope (mV/dec)	Log (i _{ex})
cv-Ni	0.25	-3.13
cc-Ni	0.34	-3.97
Commercial Ni foil	0.32	-3.72

5.4.3 Stability assessments

Stability tests were carried out by applying a constant potential (-1.0 V vs. Ag/AgCl) for a prolonged period of time on both the cc- and cv-Ni samples. As shown in Figure 5.6, the current density was very stable during the ten-hour testing period, demonstrating excellent stability. We also did a stability test involving 201 consecutive CV cycles with one cv-Ni catalyst sample. As shown in the inset of Figure 5.6, the first and 201st cycles are identical in shape as well as peak current densities, a result that also demonstrates excellent stability over a long period of time.



Figure 5.6. Current density profile of cv-Ni in a 0.1 M neutral phosphate buffer under an applied potential of -1.0V vs. Ag/AgCl. After the first two hours, the current density was steady over the remaining eight-hour testing period, demonstrating stable catalytic performance. The inset shows the cyclic voltammograms of the first and 201st cycles of a functionality test on a single cv–Ni catalytic sample. The similarity in the two profiles as well as peak current densities is an indication of very stable catalytic performance.

5.4.4 Effects of electrolyte concentration on electrodeposition

We synthesized catalysts using different concentrations of electrolyte. The number of CV cycles was tuned to ensure an identical accumulated charge of 1 C during each deposition. All samples had a similar onset potential of -0.1V vs. NHE for the hydrogen evolution reaction, but the catalytic performance varied with electrolyte concentration. Table 5.2 is a summary of the corresponding current densities at -1.2 V vs. Ag/AgCl. The results portray a typical volcano behavior, with the sample synthesized in 200 mM solution giving the best performance. In general, this profile suggests that electrolyte concentration offers another possible parameter for optimization.

Electrolyte Concentration / mM	Peak current density / (mA/cm ²)
25	0.77±0.01
50	1.22±0.16
100	1.49±0.08
200	1.83±0.04
250	1.45±0.10

Table 5.2. Peak current density at -0.5 V vs. NHE for samples synthesized in different concentrations of electrolyte.

5.4.5 Hypothesis of the formation mechanism

Although we do not yet understand the exact mechanism for the formation of the structured film, we propose a plausible hypothesis. In the traditional constant current deposition process, Ni^{2+} ions are continuously dragged to the electrode surface at a constant rate, and reduced to metallic Ni. In the CV deposition case, the applied potentials are constantly changing, therefore Ni^{2+} ions are dragged to the surface at various rates, then reduced. Also, over certain ranges of potential, the Ni^{2+} ions are not reduced, resulting in the formation of only $Ni(OH)_2$ hydrolysis species. Under this scenario, it is possible for the walnut-shaped 'core-shell' type of structures to form with metallic Ni accumulating in the center of the particles, surrounded by the nanowall-shaped $Ni(OH)_2$ hydrolysis species on the outer shell. Under such conditions, we would expect a higher Ni to $Ni(OH)_2$ ratio in the central area of the particles than on the edge. Our preliminary EDS and ID analysis data support this hypothesis, showing that the ratio of Ni to $Ni(OH)_2$ in the center is 6.74 to 1, which is about twice the ratio on the edge.

Some studies have reported that Ni(OH)₂/metal interfaces play an important role in the hydrogen evolution reaction, with the Ni(OH)₂/Ni interface being the most effective^{101,102}. Usually, such interfaces are created by complicated wet chemistry procedures. However, we have shown in this work that such interfaces can also be created using simple CV electrodeposition methods. This simpler more controllable synthesis method has successfully enabled a core-shell structure that contains large numbers of Ni(OH)₂/Ni interfaces, leading to excellent catalytic performance towards the hydrogen evolution reaction.

5.5 Conclusions

We have developed a new method of synthesizing Ni/Ni(OH)₂ HER catalysts *in situ* onto conductive surfaces via a simple electrodeposition method using cyclic voltammetry. This simple and inexpensive approach involves a single CV session of two potential ramps ranging from 0 to -1.2V. The resulting catalytic film has a uniform surface morphology consisting of an evenly distributed monolayer of spherical particles that promotes overall catalytic performance. The resulting film has stable and much better catalytic performance than traditional cc-Ni HER catalysts or Ni foil, with an effectiveness that shifts significantly towards that of platinum. We believe this work provides an easier inexpensive way of synthesizing this HER catalyst, and demonstrates the potential for using CV as a deposition method. This work also has the potential to serve as a new way of creating large quantities of Ni(OH)₂/metal interfaces to promote the performance of HER catalysts, and provide options for catalyst synthesis and optimization through electrochemistry.

CHAPTER 6. SUGGESTIONS FOR FUTURE RESEARCH

6.1 Introduction

In Chapters 3 and 5, we discussed our research activities on developing electrocatalysts for both the water oxidation and hydrogen evolution reactions. Although both the MnO_x WOC and Ni/Ni(OH)₂ HERC have excellent and stable catalytic functionalities, there are still some parameters that can be further explored to enhance performance. Also, some characterization devices used in this work are not ideal for the current system. Thus, more advanced instrumentation could be used in the future to better characterize both catalytic films. Finally, it would serve a valuable function to integrate more precise measurements and computational work to guide future electrocatalyst synthesis.

For system integration, our eventual goal was to build an affordable but highly effective solardriven water splitting device. As discussed in Chapters 1 and 4, traditional AP devices have been based on the schemes in Figure 6.1. Our solar-driven photoanode for water splitting is also based on the two-chip face-to-face scheme. Although performances of such devices could be enhanced by adopting highly effective solar harvesting systems and electrocatalysts, its overall performance is still limited by mass transfer resistance. Another big issue is that the density of solar radiation is quite low, with the result that large quantities of semiconducting materials and catalysts would be required to achieve effective hydrogen production. In this chapter, we suggest some future work for both electrocatalyst synthesis and system integration to build solar-driven water splitting devices.

Figure 6.1. Schematics of two water splitting devices. Each scheme has a photoanode complex containing a solar harvester and a WOC, and a counter electrode for hydrogen ion reconstitution to produce fuel. a. Two-chip protocol (face-to-face scheme), where the counter electrode does not share the conductive support with the photoanode; b. One-chip protocol, where the counter electrode and photoanode share one conductive support.

6.2 Further research on improving the MnO_x WOC

The dual series CV electrodeposition protocol described in Chapter 3 is a novel and excellent method for synthesizing WOCs, in comparison to other methods. Multi-series CV has tunable parameters such as number of series, potential ranges of each series, number of cycles in each series, ionic strength of the electrolyte, and the concentration of doping elements. Each parameter may have a specific effect on the resulting catalyst. It would be of scientific value to assess the effects of these parameters on the resulting catalytic films. This could also lead to a comprehensive understanding to enable rational design of catalysts.

In our current research, we used XRD to measure the oxidation state of the MnO_x catalyst. However, XRD is not a very accurate technique for measuring the oxidation state of manganese compounds. We recommend advanced x-ray absorption techniques such as x-ray absorption near edge structure (XANES) and extended x-ray absorption fine structure (EXAFS) to measure the oxidation states and the bounding patterns for MnO_x WOCs. Also, since manganese is notoriously hard for traditional transmission detectors, we would recommend the use of off-angle fluorescence detectors. This would be valuable for developing a comprehensive understanding of the mechanism for the formation of the catalytic films.

6.3 Further research on improving the Ni/Ni(OH)2 HER catalyst

The single series CV procedure described in Chapter 5 is a new method for synthesizing Ni/Ni(OH)₂ HER catalysts *in situ*. There are still other parameters that should be explored. Some, such as doping elements, may have high possibilities to promote catalytic performance or catalyst stability. We again suggest using x-ray absorption (XAS) techniques to study the details

of changes in oxidation states, and bonding patterns during electrodeposition and functional assessments. We also suggest the use of environmental SEM to characterize changes in surface morphology during the deposition. Combining these two characterization results listed above, one can get a detailed knowledge of both surface morphology and changes in oxidation states and develop a theory to guide the future development of electrocatalysts.

6.4 Further research on system integration

To address the problems in our current water splitting system, we recommend a new scheme featuring multiple photoanode modules for further investigation (Figure 6.2). Each photoanode module contains: a) a device that can collect solar energy over a large area (front surface) and concentrate it in a much smaller area along the edge; b) one or more layers of semiconducting material to convert the solar energy to the electrical potential required to power the water splitting reaction; c) a layer of conductive material such as ITO to be used as the substrate for the electrodeposition of the MnO_x WOC (to enhance the electron-hole separation, a thin hole-attraction layer could also be coated); and d) a layer of MnO_x WOC synthesized using the method described above. To minimize mass transfer resistance, we propose using the photoanode modules to form microfluidic water channels that reduce the distance for ion migration. Hydrogen evolution catalysts would be deposited on the bottom of the water channels via electrodeposition.

Figure 6.2. Overview of the proposed scheme. All the components of the proposed solar-driven water splitting device are supported by a robust conductive substrate. The squares are the AP modules which can be engineered into various shapes and sizes. Water channels are formed by the edges of the AP modules. The water oxidation catalyst is coated on the edges of the AP modules, and the hydrogen evolution catalyst is coated on the bottom of the water channel to greatly reduce the distance for ion migration, resulting in a much lower resistance to mass transfer. The detailed structure of the AP module is shown in the figure below.

Figure 6.3. Cross-sectional view of the AP module in a microfluidic channel. Each AP module features a) a solar concentrator to collect solar energy on the larger surface and transmit it to the edge; b) a convertor to transform the concentrated solar energy into the potential required for the water splitting reaction; c) a conductive layer for the electrodeposition of WOC; d) an optional p+ layer to promote electron-hole separation; and e) a WOC layer directly in contact with water.

Since the solar concentrator is a flexible material, the resulting photoanode modules can be fabricated in various sizes and shapes. As a result, the size and shape of the final device could be tuned by changing the number and pattern of the photoanode modules. Also, the final device could be attached to any surfaces if the substrate is not rigid. Successful development of a device based on this new scheme would be of great scientific and technological value.

As also described in Chapter 4, Mn-based WOCs have the capability for integration with semiconducting materials like TiO_2 . The major problem with TiO_2 is that its absorption spectrum is quite limited to the far UV range. Thus, if future researchers could find suitable dyes to broaden the absorption spectrum of the solar harvester, the efficiency of the photoanode modules could be greatly improved.

CHAPTER 7. CONCLUSIONS

This dissertation is a detailed account of our group's research activities in developing highly effective yet inexpensive electrocatalysts for the water splitting reaction. In Chapter 2, we demonstrated that a Co-OXO water oxidation catalyst that is synthesized *ex situ* can be effectively immobilized onto conductive surfaces to enhance the recyclability and reusability of the resulting catalytic film without losing functionality.

Chapter 3 covered our research on developing a dual-series cyclic voltammetry (CV) electrodeposition method to synthesize a MnO_x *in situ* on conductive surfaces. The new method involves two consecutive series of CV scans over different potential ranges, followed by calcination. Our results showed that the MnO_x WOC has stable and excellent catalytic performance over long periods of time. We also proposed a nucleation-growth formation theory that agrees with the experimental observations.

In Chapter 4, we report on our study to integrate Mn-based WOC synthesized in situ with the semiconducting material TiO₂ to build a functional photoanode. The photoanode has a double-layer structure that promotes water splitting under ultraviolet (UV) illumination. Although the current system is only functional under UV light, the addition of various dyes could broaden its absorption spectrum and lead to a more effective system.

Chapter 5 covered our study on developing effective electrocatalysts for the hydrogen evolution reaction (HER). We used only one series of CV to synthesize the Ni/Ni(OH)₂ HER catalyst in

situ onto conductive surfaces. Our results show that the catalytic film has a surface morphology of firmly packed monolayer of walnut-shaped spherical particles, with stable and excellent performance. We believe this work has provided an inexpensive way to create large numbers of Ni(OH)₂/metal interfaces to promote the performance of HER catalysts.

We believe the results of the four studies listed in this dissertation are valuable in opening new ways of synthesizing highly effective yet low cost electrocatalysts *in situ* on conductive surfaces. These studies have the potential to help guide the development of highly effective solar-driven water splitting devices in the future.

BIBLIOGRAPHY

BIBLIOGRAPHY

(1) Hammarstrom, L.; Hammes-Schiffer, S.: Artificial Photosynthesis and Solar Fuels. *Accounts Chem Res* **2009**, *42*, 1859-1860.

(2) Marcus, R. J.: Chemical Conversion of Solar Energy. *Science* **1956**, *123*, 399-405.

(3) Morton, O.: Solar energy: A new day dawning?: Silicon Valley sunrise. *Nature* **2006**, *443*, 19-22.

(4) Rosen, M.: Entropy production and exergy destruction: Part I—hierarchy of Earth's major constituencies. *International Journal of Hydrogen Energy* **2003**, *28*, 1307-1313.

(5) Crabtree, G. W.; Lewis, N. S.: Solar energy conversion. *Phys Today* **2007**, *60*, 37-42.

(6) Dennis, C.; Davey, R.; Stein, W.: Solar energy: Radiation nation. *Nature* **2006**, *443*, 23-24.

(7) Parida, B.; Iniyan, S.; Goic, R.: A review of solar photovoltaic technologies. *Renew Sust Energ Rev* **2011**, *15*, 1625-1636.

(8) Gul, M.; Kotak, Y.; Muneer, T.: Review on recent trend of solar photovoltaic technology. *Energ Explor Exploit* **2016**, *34*, 485-526.

(9) Zhang, X. X.; Zhao, X. D.; Smith, S.; Xu, J. H.; Yu, X. T.: Review of R&D progress and practical application of the solar photovoltaic/thermal (PV/T) technologies. *Renew Sust Energ Rev* **2012**, *16*, 599-617.

(10) Tachibana, Y.; Vayssieres, L.; Durrant, J. R.: Artificial photosynthesis for solar water-splitting. *Nat Photonics* **2012**, *6*, 511-518.

(11) Hou, H. J. M.; Allakhverdiev, S. I.; Najafpour, M. M.; Govindjee: Current challenges in photosynthesis: from natural to artificial. *Front Plant Sci* **2014**, *5*.

(12) Allakhverdiev, S. I.: Photosynthesis research for sustainability: From natural to artificial Preface. *Bba-Bioenergetics* **2012**, *1817*, 1107-1109.

(13) Magnuson, A.; Styring, S.: Molecular Chemistry for Solar Fuels: From Natural to Artificial Photosynthesis. *Aust J Chem* **2012**, *65*, 564-572.

(14) Evans, A.; Strezov, V.; Evans, T. J.: Assessment of sustainability indicators for renewable energy technologies. *Renew Sust Energ Rev* **2009**, *13*, 1082-1088.

(15) Handoko, A. D.; Li, K. F.; Tang, J. W.: Recent progress in artificial photosynthesis: CO2 photoreduction to valuable chemicals in a heterogeneous system. *Curr Opin Chem Eng* **2013**, *2*, 200-206.

(16) Barber, J.: Biological solar energy. *Philos T R Soc A* **2007**, *365*, 1007-1023.

(17) McConnell, I.; Li, G. H.; Brudvig, G. W.: Energy Conversion in Natural and Artificial Photosynthesis. *Chem Biol* **2010**, *17*, 434-447.

(18) Berg JM, T. J., Stryer L.: e Light Reactions of Photosynthesis. In *Biochemistry*. *5th edition*.; W H Freeman: NEW YORK, 2002.

(19) Lodish H, B. A., Zipursky SL, et al.: Section 16.3 Photosynthetic Stages and Light-Absorbing Pigments. In *Molecular Cell Biology. 4th edition.*; W. H. Freeman: New York, 2000.

(20) Berg JM, T. J., Stryer L.: Section 20.1 The Calvin Cycle Synthesizes Hexoses from Carbon Dioxide and Water. In *Biochemistry*. *5th edition*.; W H Freeman: New York, 2002.

(21) Lewis, N. S.; Nocera, D. G.: Powering the planet: Chemical challenges in solar energy utilization. *P Natl Acad Sci USA* **2007**, *104*, 20142-20142.

(22) Gray, H. B.: Mechanisms of solar-driven water splitting reactions. *Abstr Pap Am Chem S* 2011, 241.

(23) Aroutiounian, V. M.; Arakelyan, V. M.; Shahnazaryan, G. E.: Metal oxide photoelectrodes for hydrogen generation using solar radiation-driven water splitting. *Sol Energy* **2005**, *78*, 581-592.

(24) Ferreira, K. N.; Iverson, T. M.; Maghlaoui, K.; Barber, J.; Iwata, S.: Architecture of the photosynthetic oxygen-evolving center. *Science* **2004**, *303*, 1831-1838.

(25) Kanady, J. S.; Tsui, E. Y.; Day, M. W.; Agapie, T.: A Synthetic Model of the Mn3Ca Subsite of the Oxygen-Evolving Complex in Photosystem II. *Science* **2011**, *333*, 733-736.

(26) Mukherjee, S.; Stull, J. A.; Yano, J.; Stamatatos, T. C.; Pringouri, K.; Stich, T. A.; Abboud, K. A.; Britt, R. D.; Yachandra, V. K.; Christou, G.: Synthetic model of the asymmetric [Mn3CaO4] cubane core of the oxygen-evolving complex of photosystem II. *P Natl Acad Sci USA* **2012**, *109*, 2257-2262.

(27) Brimblecombe, R.; Koo, A.; Dismukes, G. C.; Swiegers, G. F.; Spiccia, L.: Solar Driven Water Oxidation by a Bioinspired Manganese Molecular Catalyst. *Journal of the American Chemical Society* **2010**, *132*, 2892-+.

(28) Ruettinger, W.; Yagi, M.; Wolf, K.; Bernasek, S.; Dismukes, G. C.: O-2 evolution from the manganese-oxo cubane core Mn4O46+: A molecular mimic of the photosynthetic water oxidation enzyme? *J Am Chem Soc* **2000**, *122*, 10353-10357.
(29) Brimblecombe, R.; Swiegers, G. F.; Dismukes, G. C.; Spiccia, L.: Sustained water oxidation photocatalysis by a bioinspired manganese cluster. *Angewandte Chemie-International Edition* **2008**, *47*, 7335-7338.

(30) Chakrabarty, R.; Bora, S. J.; Das, B. K.: Synthesis, structure, spectral and electrochemical properties, and catalytic use of cobalt(III)-Oxo cubane clusters. *Inorg Chem* **2007**, *46*, 9450-9462.

(31) Hine, F.: *Electrode processes and electrochemical engineering*; Plenum Press: New York, 1985.

(32) Gray, H. B.: Powering the planet with solar fuel. *Nat Chem* **2009**, *1*, 7-7.

(33) Styring, S.: Artificial photosynthesis for solar fuels. *Faraday Discuss* **2012**, *155*, 357-376.

(34) Walter, M. G.; Warren, E. L.; McKone, J. R.; Boettcher, S. W.; Mi, Q. X.; Santori, E. A.; Lewis, N. S.: Solar Water Splitting Cells. *Chem Rev* **2010**, *110*, 6446-6473.

(35) Mola, J.; Dinoi, C.; Sala, X.; Rodriguez, M.; Romero, I.; Parella, T.; Fontrodona, X.; Llobet, A.: A new dinuclear Ru-Hbpp based water oxidation catalyst with a trans-disposition of the Ru-OH. *Dalton T* **2011**, *40*, 3640-3646.

(36) Bozoglian, F.; Romain, S.; Ertem, M. Z.; Todorova, T. K.; Sens, C.; Mola, J.; Rodriguez, M.; Romero, I.; Benet-Buchholz, J.; Fontrodona, X.; Cramer, C. J.; Gagliardi, L.; Llobet, A.: The Ru-Hbpp Water Oxidation Catalyst. *Journal of the American Chemical Society* **2009**, *131*, 15176-15187.

(37) Romain, S.; Bozoglian, F.; Sala, X.; Llobet, A.: Oxygen-Oxygen Bond Formation by the Ru-Hbpp Water Oxidation Catalyst Occurs Solely via an Intramolecular Reaction Pathway. *Journal of the American Chemical Society* **2009**, *131*, 2768-+.

(38) Kwon, G.; Ferguson, G. A.; Heard, C. J.; Tyo, E. C.; Yin, C. R.; DeBartolo, J.; Seifert, S.; Winans, R. E.; Kropf, A. J.; Greeley, J.; Johnston, R. L.; Curtiss, L. A.; Pellin, M. J.; Vajda, S.: Size-Dependent Subnanometer Pd Cluster (Pd-4, Pd-6, and Pd-17) Water Oxidation Electrocatalysis. *Acs Nano* **2013**, *7*, 5808-5817.

(39) Yang, K. R.; Matula, A. J.; Kwon, G.; Hong, J. Y.; Sheehan, S. W.; Thomsen, J. M.; Brudvig, G. W.; Crabtree, R. H.; Tiede, D. M.; Chen, L. X.; Batista, V. S.: Solution Structures of Highly Active Molecular Ir Water-Oxidation Catalysts from Density Functional Theory Combined with High-Energy X-ray Scattering and EXAFS Spectroscopy. *Journal of the American Chemical Society* **2016**, *138*, 5511-5514.

(40) Materna, K. L.; Rudshteyn, B.; Brennan, B. J.; Kane, M. H.; Bloomfield, A. J.; Huang, D. L.; Shopov, D. Y.; Batista, V. S.; Crabtree, R. H.; Brudvig, G. W.: Heterogenized Iridium Water-Oxidation Catalyst from a Silatrane Precursor. *Acs Catal* **2016**, *6*, 5371-5377.

(41) Huang, D. L.; Beltran-Suito, R.; Thomsen, J. M.; Hashmi, S. M.; Materna, K. L.; Sheehan, S. W.; Mercado, B. Q.; Brudvig, G. W.; Crabtree, R. H.: New Ir Bis-Carbonyl Precursor for Water Oxidation Catalysis. *Inorg Chem* **2016**, *55*, 2427-2435.

(42) Kanan, M. W.; Nocera, D. G.: In situ formation of an oxygen-evolving catalyst in neutral water containing phosphate and Co2+. *Science* **2008**, *321*, 1072-1075.

(43) Yuan, H.; Newton, D. L.; Seymour, L. A.; Metz, A.; Cropek, D.; Holder, A. A.; Ofoli, R. Y.: Characterization and functional assessment of a cobalt(III)-oxo cubane cluster water oxidation catalyst immobilized on ITO. *Catal Commun* **2014**, *56*, 76-80.

(44) Jiao, F.; Frei, H.: Nanostructured manganese oxide clusters supported on mesoporous silica as efficient oxygen-evolving catalysts. *Chem Commun* **2010**, *46*, 2920-2922.

(45) Pace, R. J.; Stranger, R.; Petrie, S.: Why nature chose Mn for the water oxidase in Photosystem II. *Dalton T* **2012**, *41*, 7179-7189.

(46) Yachandra, V. K.; Sauer, K.; Klein, M. P.: Manganese cluster in photosynthesis: Where plants oxidize water to dioxygen. *Chem Rev* **1996**, *96*, 2927-2950.

(47) Javed, Q.; Wang, F. P.; Rafique, M. Y.; Toufiq, A. M.; Li, Q. S.; Mahmood, H.; Khan, W.: Diameter-controlled synthesis of alpha-Mn2O3 nanorods and nanowires with enhanced surface morphology and optical properties. *Nanotechnology* **2012**, *23*.

(48) Wiechen, M.; Zaharieva, I.; Dau, H.; Kurz, P.: Layered manganese oxides for water-oxidation: alkaline earth cations influence catalytic activity in a photosystem II-like fashion. *Chem Sci* **2012**, *3*, 2330-2339.

(49) Zhou, F. L.; Izgorodin, A.; Hocking, R. K.; Spiccia, L.; MacFarlane, D. R.: Electrodeposited MnOx Films from Ionic Liquid for Electrocatalytic Water Oxidation. *Adv Energy Mater* **2012**, *2*, 1013-1021.

(50) Gorlin, Y.; Lassalle-Kaiser, B.; Benck, J. D.; Gul, S.; Webb, S. M.; Yachandra, V. K.; Yano, J.; Jaramillo, T. F.: In Situ X-ray Absorption Spectroscopy Investigation of a Bifunctional Manganese Oxide Catalyst with High Activity for Electrochemical Water Oxidation and Oxygen Reduction. *J Am Chem Soc* **2013**, *135*, 8525-8534.

(51) Indra, A.; Menezes, P. W.; Zaharieva, I.; Baktash, E.; Pfrommer, J.; Schwarze, M.; Dau, H.; Driess, M.: Active Mixed-Valent MnOx Water Oxidation Catalysts through Partial Oxidation (Corrosion) of Nanostructured MnO Particles. *Angewandte Chemie-International Edition* **2013**, *52*, 13206-13210.

(52) Pang, S. C.; Anderson, M. A.; Chapman, T. W.: Novel electrode materials for thin-film ultracapacitors: Comparison of electrochemical properties of sol-gel-derived and electrodeposited manganese dioxide. *J Electrochem Soc* **2000**, *147*, 444-450.

(53) Pinaud, B. A.; Chen, Z. B.; Abram, D. N.; Jaramillo, T. F.: Thin Films of Sodium Birnessite-Type MnO2: Optical Properties, Electronic Band Structure, and Solar Photoelectrochemistry. *J Phys Chem C* **2011**, *115*, 11830-11838.

(54) Zaharieva, I.; Chernev, P.; Risch, M.; Klingan, K.; Kohlhoff, M.; Fischer, A.; Dau, H.: Electrosynthesis, functional, and structural characterization of a water-oxidizing manganese oxide. *Energ Environ Sci* **2012**, *5*, 7081-7089.

(55) Yuan, H.; Lunt, R. R.; Blanchard, G. J.; Ofoli, R. Y.: Synthesis of MnOx Water Oxidation Catalyst on Fluorine-Doped Tin Oxide with a Dual-Series Cyclic Voltammetry Method. *Chemelectrochem* **2016**, *3*, 709-712.

(56) Gorlin, Y.; Jaramillo, T. F.: Investigation of Surface Oxidation Processes on Manganese Oxide Electrocatalysts Using Electrochemical Methods and Ex Situ X-ray Photoelectron Spectroscopy. *J Electrochem Soc* **2012**, *159*, H782-H786.

(57) Izgorodin, A.; Hocking, R.; Winther-Jensen, O.; Hilder, M.; Winther-Jensen, B.; MacFarlane, D. R.: Phosphorylated manganese oxide electrodeposited from ionic liquid as a stable, high efficiency water oxidation catalyst. *Catal Today* **2013**, *200*, 36-40.

(58) Takashima, T.; Hashimoto, K.; Nakamura, R.: Mechanisms of pH-Dependent Activity for Water Oxidation to Molecular Oxygen by MnO2 Electrocatalyst. *Journal of the American Chemical Society* **2012**, *134*, 1519-1527.

(59) Gong, J. W.; Liang, J.; Sumathy, K.: Review on dye-sensitized solar cells (DSSCs): Fundamental concepts and novel materials. *Renew Sust Energ Rev* **2012**, *16*, 5848-5860.

(60) Cherrington, R.; Hughes, D. J.; Senthilarasu, S.; Goodship, V.: Inkjet-Printed TiO2 Nanoparticles from Aqueous Solutions for Dye-Sensitized Solar Cells (DSSCs). *Energy Technol-Ger* **2015**, *3*, 866-870.

(61) Bella, F.; Gerbaldi, C.; Barolo, C.; Gratzel, M.: Aqueous dye-sensitized solar cells. *Chem Soc Rev* **2015**, *44*, 3431-3473.

(62) Li, B.; Wang, L. D.; Kang, B. N.; Wang, P.; Qiu, Y.: Review of recent progress in solid-state dye-sensitized solar cells. *Sol Energ Mat Sol C* **2006**, *90*, 549-573.

(63) Mao, X. L.; Zhou, R.; Zhang, S. W.; Ding, L. P.; Wan, L.; Qin, S. X.; Chen, Z. S.; Xu, J. Z.; Miao, S. D.: High Efficiency Dye-sensitized Solar Cells Constructed with Composites of TiO2 and the Hot-bubbling Synthesized Ultra-Small SnO2 Nanocrystals. *Sci Rep-Uk* **2016**, *6*.

(64) Gopinath, K.; Kumaraguru, S.; Bhakyaraj, K.; Thirumal, S.; Arumugam, A.: Ecofriendly synthesis of TiO2, Au and Pt doped TiO2 nanoparticles for dye sensitized solar cell applications and evaluation of toxicity. *Superlattice Microst* **2016**, *92*, 100-110.

(65) Roose, B.; Pathak, S.; Steiner, U.: Doping of TiO2 for sensitized solar cells. *Chem Soc Rev* **2015**, *44*, 8326-8349.

(66) Dette, C.; Perez-Osorio, M. A.; Kley, C. S.; Punke, P.; Patrick, C. E.; Jacobson, P.; Giustino, F.; Jung, S. J.; Kern, K.: TiO2 Anatase with a Bandgap in the Visible Region. *Nano Lett* **2014**, *14*, 6533-6538.

(67) Khan, M. M.; Ansari, S. A.; Pradhan, D.; Ansari, M. O.; Han, D. H.; Lee, J.; Cho, M. H.: Band gap engineered TiO2 nanoparticles for visible light induced photoelectrochemical and photocatalytic studies. *J Mater Chem A* **2014**, *2*, 637-644.

(68) Ni, M.; Leung, M. K. H.; Leung, D. Y. C.; Sumathy, K.: A review and recent developments in photocatalytic water-splitting using TiO2 for hydrogen production. *Renew Sust Energ Rev* **2007**, *11*, 401-425.

(69) Fujishima, A.; Honda, K.: Electrochemical Photolysis of Water at a Semiconductor Electrode. *Nature* **1972**, *238*, 37-+.

(70) Pleskov, Y. V.: A. Fujishima, K. Hashimoto, and T. Watanabe, TiO2 photocatalysis: Fundamentals and applications, Tokyo: Bks, Inc., 1999. *Russ J Electrochem*+**1999**, *35*, 1137-1138.

Bledowski, M.; Wang, L. D.; Ramakrishnan, A.; Betard, A.; Khavryuchenko, O.
V.; Beranek, R.: Visible-Light Photooxidation of Water to Oxygen at Hybrid TiO2-Polyheptazine Photoanodes with Photodeposited Co-Pi (CoOx) Cocatalyst. *Chemphyschem* 2012, *13*, 3018-3024.

(72) Tatar, D.; Duzgun, B.: The relationship between the doping levels and some physical properties of SnO2:F thin films spray-deposited on optical glass. *Pramana-J Phys* **2012**, *79*, 137-150.

(73) Mori, K.: Photo-Functionalized Materials Using Nanoparticles: Photocatalysis. J. Soc. Powder Technology **2004**, 41, 6.

(74) Uner; Graetzel; Sacilotti; Peter; Styring; Hammarstrom; Perutz; Gust; Benniston; Harriman; Burrows; Ghiggino; Guldi; Cogdell; Uner; Inoue; Faunce; Mallouk; Reisner; Szalai; Durrant; Sun; Fielden; Fujita; Campagna; Farras; Kim; Bai; He; Amao; Fukuzumi: Artificial leaf device for solar fuel production. *Faraday Discuss* **2012**, *155*, 297-308.

(75) Reece, S. Y.; Hamel, J. A.; Sung, K.; Jarvi, T. D.; Esswein, A. J.; Pijpers, J. J. H.; Nocera, D. G.: Wireless Solar Water Splitting Using Silicon-Based Semiconductors and Earth-Abundant Catalysts. *Science* **2011**, *334*, 645-648.

(76) Nocera, D. G.: The Artificial Leaf. Accounts Chem Res 2012, 45, 767-776.

(77) Osterloh, F. E.: Inorganic nanostructures for photoelectrochemical and photocatalytic water splitting. *Chem Soc Rev* **2013**, *42*, 2294-2320.

(78) Krstajic, N. V.; Jovic, V. D.; Gajic-Krstajic, L.; Jovic, B. M.; Antozzi, A. L.; Martelli, G. N.: Electrodeposition of Ni-Mo alloy coatings and their characterization as cathodes for hydrogen evolution in sodium hydroxide solution. *International Journal of Hydrogen Energy* **2008**, *33*, 3676-3687.

(79) Lee, J. K.; Yi, Y.; Lee, H. J.; Uhm, S.; Lee, J.: Electrocatalytic activity of Ni nanowires prepared by galvanic electrodeposition for hydrogen evolution reaction. *Catal Today* **2009**, *146*, 188-191.

(80) Mattarozzi, L.; Cattarin, S.; Comisso, N.; Gambirasi, A.; Guerriero, P.; Musiani, M.; Vazquez-Gomez, L.; Verlato, E.: Hydrogen evolution assisted electrodeposition of porous Cu-Ni alloy electrodes and their use for nitrate reduction in alkali. *Electrochim Acta* **2014**, *140*, 337-344.

(81) Michaux, K. E.; Gambardella, A. A.; Alibabaei, L.; Ashford, D. L.; Sherman, B. D.; Binstead, R. A.; Meyer, T. J.; Murray, R. W.: Visible Photoelectrochemical Water Splitting Based on a Ru(II) Polypyridyl Chromophore and Iridium Oxide Nanoparticle Catalyst. *J Phys Chem C* **2015**, *119*, 17023-17027.

(82) Bus, E.; van Bokhoven, J. A.: Hydrogen chemisorption on supported platinum, gold, and platinum-gold-alloy catalysts. *Phys Chem Chem Phys* **2007**, *9*, 2894-2902.

(83) Vesborg, P. C. K.; Seger, B.; Chorkendorff, I.: Recent Development in Hydrogen Evolution Reaction Catalysts and Their Practical Implementation. *J Phys Chem Lett* **2015**, *6*, 951-957.

(84) Zeng, M.; Li, Y. G.: Recent advances in heterogeneous electrocatalysts for the hydrogen evolution reaction. *J Mater Chem A* **2015**, *3*, 14942-14962.

(85) Aaboubi, O.; Ali-Omar, A. Y.; Dzoyem, E.; Marthe, J.; Boudifa, M.: Ni-Mn based alloys as versatile catalyst for different electrochemical reactions. *J Power Sources* **2014**, *269*, 597-607.

(86) Han, A.; Jin, S.; Chen, H. L.; Ji, H. X.; Sun, Z. J.; Du, P. W.: A robust hydrogen evolution catalyst based on crystalline nickel phosphide nanoflakes on three-dimensional graphene/nickel foam: high performance for electrocatalytic hydrogen production from pH 0-14. *J Mater Chem A* **2015**, *3*, 1941-1946.

(87) Pan, Y.; Liu, Y. R.; Zhao, J. C.; Yang, K.; Liang, J. L.; Liu, D. D.; Hu, W. H.; Liu, D. P.; Liu, Y. Q.; Liu, C. G.: Monodispersed nickel phosphide nanocrystals with different phases: synthesis, characterization and electrocatalytic properties for hydrogen evolution. *J Mater Chem A* **2015**, *3*, 1656-1665.

(88) Popczun, E. J.; McKone, J. R.; Read, C. G.; Biacchi, A. J.; Wiltrout, A. M.; Lewis, N. S.; Schaak, R. E.: Nanostructured Nickel Phosphide as an Electrocatalyst for the Hydrogen Evolution Reaction. *J Am Chem Soc* **2013**, *135*, 9267-9270.

(89) Zhang, L.; Xiong, K.; Chen, S. G.; Li, L.; Deng, Z. H.; Wei, Z. D.: In situ growth of ruthenium oxide-nickel oxide nanorod arrays on nickel foam as a binder-free integrated cathode for hydrogen evolution. *J Power Sources* **2015**, *274*, 114-120.

(90) Zou, X. X.; Zhang, Y.: Noble metal-free hydrogen evolution catalysts for water splitting. *Chem Soc Rev* **2015**, *44*, 5148-5180.

(91) Tang, C.; Xie, L. S.; Sun, X. P.; Asiri, A. M.; He, Y. Q.: Highly efficient electrochemical hydrogen evolution based on nickel diselenide nanowall film. *Nanotechnology* **2016**, *27*.

(92) Qian, X.; Hang, T.; Shanmugam, S.; Li, M.: Decoration of Micro-/Nanoscale Noble Metal Particles on 3D Porous Nickel Using Electrodeposition Technique as Electrocatalyst for Hydrogen Evolution Reaction in Alkaline Electrolyte. *Acs Appl Mater Inter* **2015**, *7*, 15716-15725.

(93) Yang, L. B.; Qi, H. L.; Zhang, C. X.; Sun, X. P.: An efficient bifunctional electrocatalyst for water splitting based on cobalt phosphide. *Nanotechnology* **2016**, *27*.

(94) Institute, N.: *Nickel Plating Handbook*, 2014.

(95) Gong, M.; Wang, D. Y.; Chen, C. C.; Hwang, B. J.; Dai, H. J.: A mini review on nickel-based electrocatalysts for alkaline hydrogen evolution reaction. *Nano Res* **2016**, *9*, 28-46.

(96) Lu, Z. Y.; Chang, Z.; Zhu, W.; Sun, X. M.: Beta-phased Ni(OH)(2) nanowall film with reversible capacitance higher than theoretical Faradic capacitance. *Chem Commun* **2011**, *47*, 9651-9653.

(97) Li, G. H.; Wang, X. W.; Liu, L.; Liu, R.; Shen, F. P.; Cui, Z.; Chen, W.; Zhang, T.: Controllable Synthesis of 3D Ni(OH)(2) and NiO Nanowalls on Various Substrates for High-Performance Nanosensors. *Small* **2015**, *11*, 731-739.

(98) Li, J. T.; Zhao, W.; Huang, F. Q.; Manivannan, A.; Wu, N. Q.: Single-crystalline Ni(OH)(2) and NiO nanoplatelet arrays as supercapacitor electrodes. *Nanoscale* **2011**, *3*, 5103-5109.

(99) Yuan, F. F.; Ni, Y. H.; Zhang, L.; Yuan, S. M.; Wei, J. D.: Synthesis, properties and applications of flowerlike Ni-NiO composite microstructures. *J Mater Chem A* **2013**, *1*, 8438-8444.

(100) Kim, S. I.; Lee, J. S.; Ahn, H. J.; Song, H. K.; Jang, J. H.: Facile Route to an Efficient NiO Supercapacitor with a Three-Dimensional Nanonetwork Morphology. *Acs Appl Mater Inter* **2013**, *5*, 1596-1603.

(101) Subbaraman, R.; Tripkovic, D.; Strmcnik, D.; Chang, K. C.; Uchimura, M.; Paulikas, A. P.; Stamenkovic, V.; Markovic, N. M.: Enhancing Hydrogen Evolution Activity in Water Splitting by Tailoring Li+-Ni(OH)(2)-Pt Interfaces. *Science* **2011**, *334*, 1256-1260.

(102) Danilovic, N.; Subbaraman, R.; Strmcnik, D.; Chang, K. C.; Paulikas, A. P.; Stamenkovic, V. R.; Markovic, N. M.: Enhancing the Alkaline Hydrogen Evolution Reaction Activity through the Bifunctionality of Ni(OH)2/Metal Catalysts. *Angew Chem Int Edit* **2012**, *51*, 12495-12498.